STATE AID LAW OF THE EUROPEAN UNION

EDITED BY HERWIG C. H. HOFMANN CLAIRE MICHEAU



STATE AID LAW OF THE EUROPEAN UNION

State Aid Law of the European Union

Edited by HERWIG C. H. HOFMANN

Professor of European and Transnational Public Law University of Luxembourg

CLAIRE MICHEAU

Policy Officer, European Commission Associate Professor, University of Helsinki

With the Support of BUCURA C. MIHAESCU-EVANS University of Luxembourg





Great Clarendon Street, Oxford, OX2 6DP, United Kingdom

Oxford University Press is a department of the University of Oxford. It furthers the University's objective of excellence in research, scholarship, and education by publishing worldwide. Oxford is a registered trade mark of Oxford University Press in the UK and in certain other countries

© The several contributors 2016

The moral rights of the authors have been asserted

First Edition published in 2016 Impression: 1

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted, in any form or by any means, without the prior permission in writing of Oxford University Press, or as expressly permitted by law, by licence or under terms agreed with the appropriate reprographics rights organization. Enquiries concerning reproduction outside the scope of the above should be sent to the Rights Department, Oxford University Press, at the address above

You must not circulate this work in any other form and you must impose this same condition on any acquirer

Crown copyright material is reproduced under Class Licence Number C01P0000148 with the permission of OPSI and the Queen's Printer for Scotland

Published in the United States of America by Oxford University Press 198 Madison Avenue, New York, NY 10016, United States of America

British Library Cataloguing in Publication Data

Data available

Library of Congress Control Number: 2015960804

ISBN 978-0-19-872746-0

Printed and bound by CPI Group (UK) Ltd, Croydon, CR0 4YY

Links to third party websites are provided by Oxford in good faith and for information only. Oxford disclaims any responsibility for the materials contained in any third party website referenced in this work.

Disclaimer: Any opinions expressed by the editors and authors are strictly personal and do not reflect the views of any institution or organization.

Foreword

State Aid is one of the sectors in which the EU legal order seeks to find a balance between Competition Law and Member States' support policies to undertakings to achieve a well-functioning and equitable economy. Identifying this balance is one of the most complicated challenges for the EU, which, in this field, needs to combine free market principles and Member States' interventions. Furthermore, over the years, State Aid control provisions have evolved, stemming from the Treaty, secondary legislation, as well as Court rulings. The result of this regulatory and jurisprudential stratification is the increasing complexity of State Aid Law. As a consequence, there are some recurring issues and also new ones that arise. What is State Aid? When is State Aid de jure compatible with EU Law? When is State Aid compatible after a discretionary assessment? What State Aid is exempt from notification? What are the features of the administrative procedures before the Commission? What are the principles stemming from the Court of Justice's case law? What is the relationship between EU State Aid and the World Trade Organization (WTO)? Faced with all those questions, State Aid Law of the European Union edited by Herwig Hofmann and Claire Micheau is an absolute must-read book for four reasons.

First, the book has an interdisciplinary approach. State Aid cannot be completely understood without examining the interplay between both the political and economic motivations that guide states and their sub-structures. Thus, the book has the virtue of dealing with the political, economic, legal, and institutional dimensions of State Aid, by treating the matter in an interdisciplinary and multi-faceted, scholarly way. The added value of this approach is clear: instead of valuable insights from each discipline, the book offers a consolidated approach, which combines all these different vantage points. Part I of the book is about these different dimensions. It outlines the existing debates and gives new insight into academic writings that lead the reader to an understanding of the reasons for and conditions of controlling State Aid in a multilevel structure. The value of this is to show that competition law is not the only perspective to understanding the evolution of State Aid control.

Second, as well as this interdisciplinary approach, the book covers every field of 'traditional' State Aid in the European Union. Indeed, Parts II–V explore in detail the key elements of the subject.

Part II deals with the notion of State Aid. It examines the criterion of State Origin, the notion of undertaking, the criterion of advantage and selectivity, the distortion of competition, and the effect on trade between EU Member States. It also contains two important additions to the discussion of the notion of aid, that is, the use of concepts of public procurement and of European Funds in State Aid control. Part II seeks to, and does successfully, explain the notion of State Aid in light of the recent modernization of the State Aid programme and, in particular, the chapters on public procurement and European Funds give the full picture of this notion.

Part III concerns Compatibility rules on State Aid. It visits the big questions, such as the relation between policy-specific regulations and rules on State Aid, and gives an in-depth analysis of the conditions of compatibility in the various sub-categories created by Union law. Thus, *de jure* compatible Aid under Article 107(2) TFEU and aid subject to a discretionary assessment under Article 107(3) TFEU are examined, as well as Transport Aid, which is assessed on the basis of Article 93 TFEU. Moreover, this part looks at State Aid that is exempt from notification to the Commission according to the General Block Exemption

viii Foreword

Regulation (GBER). Therefore, Part III gives a thorough overview of the set of problems, contributes to the debate, and proposes original solutions to the different issues.

Parts IV and V deal with the fundamental and different roles of two key institutional players in State Aid control: the Commission and the Court of Justice. In particular, Part IV discusses procedures before the Commission. However, it is not a simple commentary on these procedures, but rather it examines a number of basic principles of the Commission's administrative procedures in relation to State Aid investigations and focuses on the limited rights of interested parties in these procedures. On that point, Part IV offers several explanations following three headings: the addressee factor, the dual role of the Commission and national courts, and the so-called 'one size fits all' approach. It also comments on the developments in case law, Commission practice, and avenues available at the national level that may alleviate this situation. Part V contains a thorough examination of the case law of the Court of Justice in the field of State Aid, in the light of the basic constitutional features of the EU system of judicial review. Thus, the allocation of powers between Union and national courts in the field of EU State Aid, the remedies available at centralized and decentralized levels, as well as their interactions are examined, taking into account the fact that the EU judicial system follows a dual approach based on mutual completion of remedies granted at both Union and Member State level. Both neophytes and skilled practitioners of EU State Aids will find Parts II to V extremely useful.

Third, the book has a global dimension. Indeed, after having examined State Aid in the EU legal order, Part VI focuses on the analysis of two sets of instruments: WTO law on subsidies established in the context of the General Agreement on Tariffs and Trade (GATT) and the Agreement on Subsidies and Countervailing Measures (SCM Agreement) and EU trade defence instruments. The first chapter of this part compares the WTO subsidy laws to the notions of State Aid in EU law. This comparative approach presents a more detailed understanding of the EU State Aid regime by contrasting it with WTO law, and is useful in providing concrete reasons and potential solutions for conflicts of law. Indeed, it comes to the interesting conclusion that both systems can learn from each other. The second chapter looks at the EU's anti-subsidy instruments. It points out that since 1995 anti-subsidy instruments have progressively become the key tool through which the EU has addressed subsidization by trading partners.

Fourth, the composition of the team of authors—judges at the Court of Justice and the General Court, legal secretaries of the latter, officials of the European Commission, skilled lawyers, and distinguished scholars—guarantees a perfect combination of leading academic and practical expertise. Thus, the theoretical approach and real-world practice blend in a book that challenges present issues on EU State Aid and proposes original solutions.

For all those reasons, scholars and practitioners before the Court of Justice should read this book, which is a key tool to understanding the evolution of State Aid, to explore elements of potential answers to the most common, albeit complex, questions and to conceive possible future developments on the subject.

Marc Jaeger President of the General Court, Court of Justice of the European Union

Summary Table of Contents

Ta Li	able of Cases able of Legislation st of Abbreviations st of Contributors	xxiii xlv lxi lxiii
	PART I RATIONALES FOR STATE AID RULES	
1.	State Aid Review in a Multi-level System: Motivations for Aid, Why Control It, and the Evolution of State Aid Law in the EU <i>Herwig C H Hofmann</i>	3
2.	Evolution of Aid in the EU: Classifying Different Types of Countries, and the Financial and Economic Crisis <i>Raj Chari</i>	12
3.	Evolution of State Aid Rules: Conceptions, Challenges, and Outcomes <i>Claire Micheau</i>	18
4.	The Legal Framework to Subsidies and State Aid Review Herwig C H Hofmann	36
	PART II THE NOTION OF STATE AID	
1.	Criterion of State Origin Erika Szyszczak	65
2.	The Notion of Undertaking Wolf Sauter	74
3.	Criterion of Advantage Aindrias Ó Caoimh and Wolf Sauter	84
4.	Criterion of Selectivity Roberto Cisotta	129
5.	Distortion of Competition and Effect on Trade between EU Member States Erika Szyszczak	151
6.	Public Procurement and State Aid Christopher Bovis	161
7.	European Structural and Investment Funds and State Aid Control: Two Policies at Odds with Each Other? Katerina Pantazatou	187
	PART III RULES FOR THE COMPATIBILITY OF STATE AID)
1.	General Theory on Compatibility of State Aid <i>Tim Maxian Rusche</i>	223
2.	De Jure Compatible Aid under Article 107(2) TFEU Massimo Francesco Orzan	234
3.	Aid Subject to a Discretionary Assessment under Article 107(3) TFEU Ramona Ianus and Massimo Francesco Orzan	240

4.	Transport Aid Tim Maxian Rusche	308
5.	Aid Exempted from Notification to the Commission: The General Block Exemption Regulation (GBER) Ramona Ianus	324
	PART IV PROCEDURES BEFORE THE COMMISSION	
1.	The Administrative Procedure—the Privileged Dialogue Leigh Hancher	341
2.	The Different Stages in the State Aid Procedure Leigh Hancher, Francesco Maria Salerno, and Michael Schütte	348
3.	Recovery of Unlawful Aid and the Role of National Courts Bucura C Mihaescu-Evans	381
	PART V JUDICIAL REVIEW BEFORE THE COURT OF JUSTICE OF THE EUROPEAN UNION	
1.	Introduction: EU State Aid Litigation in a Quasi-Federal System <i>Hanns Peter Nehl</i>	401
2.	Direct Actions and Judicial Review before the Union Courts Hanns Peter Nehl	406
3.	Decentralized Judicial Review and Enforcement of EU State Aid Rules <i>Viktor Kreuschitz</i>	450
	PART VI GLOBAL APPROACH OF STATE AID LAW—WTO LAW AND EU TRADE DEFENCE INSTRUMENTS	
1.	WTO Subsidy Laws: The International Regulation of State Aid <i>Luca Rubini</i>	469
2.	Anti-subsidy Law and Practice of the European Union Juhi Sud and Edwin Vermulst	508
In	dex	565

Detailed Table of Contents

Table of Cases	XX111
Table of Legislation	xlv
List of Abbreviations	lxi
List of Contributors	lxiii
PART I RATIONALES FOR STATE AID RULES	
1. State Aid Review in a Multi-level System: Motivations for Aid,	
Why Control It, and the Evolution of State Aid Law in the EU	3
Section 1. Objectives of Aid and States' Motivations for Giving	3
Section 2. Why Control State Aid and Subsidies?	5
A. Meta-regulation of national spending policies	5
B. State aid control as competition law	6
C. State aid control as contribution to integration through the creation	0
and maintenance of the internal market	9
D. State aid control as control of public spending	10
2. Evolution of Aid in the EU: Classifying Different Types of Countries,	
and the Financial and Economic Crisis	12
3. Evolution of State Aid Rules: Conceptions, Challenges, and Outcomes	18
Introduction	18
Section 1. The Early Negotiations to Conceive the Notion of State Aid	18
Section 2. The Pioneering Role Played by the Court of Justice	21
Section 3. The Role of the Commission as a Rule-maker	23
A. Development of regulatory texts	24
B. The soft law as informal rule-making	25
C. The distinctive importance of the Procedural Regulation	28
Section 4. Reforms Over the Last Decade: The Shifting Landscape	
of State Aid Rules	30
A. The State Aid Action Plan: Objectives and outcomes	31
B. The State Aid Modernization initiative	33
Conclusion	35
4. The Legal Framework to Subsidies and State Aid Review	36
Section 1. WTO Law	36
Section 2. EU Law	38
A. EU primary law—treaties and general principles of law	39
1. State aid law in the context of the Treaties	39
2. General principles of EU law and procedural rights	41
2.1. Good administration, rights of defence, and participation rights	41
2.2. Right to an effective legal remedy and State aid	46
B. Derived law and delegation of powers	47 48
 Legislation Group exemption regulations 	48
3. Administrative guidelines	49
4. Sector investigations and reports	52
5. Agreements as forms of administrative rule-making	53
6. Single-case decision-making	54

	C. The extent of the delegation of powers: Discretion and its limits	55
	1. Definition and consequences	55
	2. Limitations of discretion	58
	2.1. The expansion of the notion of facts	58
	2.2. Duty of care	59
	2.3. Self-limitation of discretion by the Commission	60
	2.4. Limitation of Commission discretion in control of Member State action	61
	Section 3. Member State Law	61
	Conclusion	62
	PART II THE NOTION OF STATE AID	
1.	Criterion of State Origin	65
	Section 1. Direct or Indirect Granting through State Resources	65
	Section 2. Imputability of a Measure to the State	70
	Conclusion	73
_		
2.	The Notion of Undertaking	74
	Section 1. Notion of Undertaking in Competition law	74
	The key notion of economic activity	75
	Section 2. Sector-based Notion of Undertaking	76
	A. Public policy	76
	B. Social security	77
	C. Healthcare	79
	D. Education and research	80
	E. Infrastructures	81
	Conclusion	83
3.	Criterion of Advantage	84
	Section 1. Notion of Advantage in State Aid Law	84
	Section 2. Assessing the Advantage in State Aid: <i>Altmark</i> Case Law	
	and Services of General Economic Interest (SGEI)	86
	A. Advantage in the context of SGEI	87
	1. What are SGEI and what is their purpose?	87
	2. What is the legal basis for SGEI?	88
	3. Which services are SGEI?	90
	4. Who designates SGEI?	91
	5. Why are SGEI State aid relevant?	92
	B. The relevance of the <i>Altmark</i> case law	92
	1. The early case law: Pre- <i>Altmark</i> cases and the <i>Altmark</i> case	92
	2. Relationship between Article 107(1) and Article 106(2) TFEU	94
	3. The Altmark Package Mark I	95
	4. The Altmark Package Mark II	96
	5. Summary of the system of review of State aid	0.0
	and public service compensation	98
	5.1. Transport sector and public services:	100
	Overview of Regulation 1370/2007	100 101
	C. Sector-based examples 1. SGEI in the context of healthcare	101
	2. SGEI in the context of broadband	101
	D. Conclusion	103
	Section 3. The Market Economy Operator Test	104
	A. Historical origins	105
	11. 11000110MI OIIMIIIO	100

	Detailed Table of Contents	xiii
	B. Identification of the hypothetical market operator 'in normal market conditions' C. Private vendor	115 122
	D. The nature of judicial review of Commission decisions	126
	E. Economic appraisal Conclusion	127 127
4		
4.	Criterion of Selectivity	129
	Section 1. Selectivity as a General Canon	129
	Section 2. The Discretion of National Authorities	131
	Section 3. Selectivity in Tax Matters A. Tax law and selectivity: Some general issues	132 132
	B. Case study of the three-step (derogation) test as regards	132
	the different types of selectivity	133
	1. Material selectivity (in particular, in fiscal matters)	134
	2. The elaboration and application of the derogation test: Definition of	
	the reference system and assessment of the treatment of	125
	the undertakings concerned 3. Justification by the logic of the tax system	135 139
	3.1. The inherent logic of the system and the external objectives:	137
	Evaluation of a derogation from the application of	
	the general tax system	139
	3.2. Practical applications of the justification by the logic of the tax system	140
	3.3. Geographic selectivity	144
	4. Selectivity between policy choices and economic analysis4.1. Comparing undertakings' position in the light of the objectives	146
	of a tax measure	146
	4.2. The coherence of national legislations: Comparison with	
	the freedoms of movement case law	147
	Conclusion	150
5.	Distortion of Competition and Effect on Trade between EU Member States	151
	Section 1. Distortion of Competition	153
	Section 2. Effect on Trade between the EU Member States	154
	Section 3. <i>De Minimis</i> Rules	158
	Conclusion	160
6.	Public Procurement and State Aid	161
	Introduction	161
	Section 1. Common Themes in State Aid and Public Procurement A. The State through the prism of State aid <i>acquis</i>	163
	and public procurement regulation	163
	Section 2. Financial Engagement between Public and Private Sectors A. The notion of transferring resources through the prism of State aid <i>acquis</i>	166 166
	B. The notion of public contracts through public procurement regulation	167
	Section 3. Public Services as the Ground Where State Aid Meets	207
	Public Procurement	168
	A. The notion of public services	168
	B. The treatment of financing the delivery of public services in the EU	172
	Section 4. The Root of Exclusivity in State Aid and Public Procurement	180
	A. The ordo-liberal nature of public procurement regulation 1. Public procurement as part of the European Integration	180 180
	2. Contract compliance and public procurement	182
	2.1. A rule of reason	183
	Conclusion	185

/.	European Structural and Investment Funds and State Aid Control:	
	Two Policies at Odds with Each Other?	187
	Introduction	187
	Section 1. EU Cohesion Policy and ESI Funds: Objectives and Complex	
	Administrative Structures	188
	Section 2. State Aid Control and ESI Funds: Concepts and Objectives	190
	A. Competition policy and cohesion policy: Clash or complementarity?	191
	B. The new legal framework: Is consistency achieved?	195
	Section 3. The New Legal Framework for ESI Funds	197
	A. <i>Ex ante</i> conditionalities, national administrative capacity building,	
	and State aids	198
	B. Financial instruments in the ESI Funds and compliance with State aids	199
	C. Alignment with the SAM objectives	200
	Section 4. The Notion of State Aid and ESI Funds	203
	A. State resources and imputability to the State: Margin of discretion of	
	the Member States in the implementation process	204
	B. Implications of the classification of ESI Funds as 'State aids':	
	Aid intensities and cumulation rules	209
	Section 5. WTO and Union Subsidization	211
	A. CAP and export subsidies	211
	B. The Agreement on Subsidies and Countervailing Measures and	
	the Agriculture Agreement (AG)	212
	C. WTO ruling on sugar and its implications	213
	Conclusion	216
	PART III RULES FOR THE COMPATIBILITY OF STATE AID	
1.	General Theory on Compatibility of State Aid	223
	Section 1. On the Distribution of Powers	223
	Section 2. On the Burden of Proof for Compatibility	224
	Section 3. On the Broad Discretion of the Commission and its Limits	224
	Section 4. On the Application of Rules on Compatibility <i>Ratione Temporis</i>	227
	Section 5. On the Assessment of State Aid Directly on the Basis	
	of Article 107(3)(c) TFEU	228
	Section 6. On the Discretion of the Commission under Article 106(2) TFEU	229
	Section 7. On the Relationship between Regulation and State Aid	230
	Section 8. On the Relationship with Other Rules of the Treaty	230
	Section 9. On the Role of Economic Analysis and Reform and Modernization	231
	•	201
2.	De Jure Compatible Aid under Article 107(2) TFEU	234
	Introduction	234
	Section 1. Aid of a Social Character Granted to Individual Consumers	235
	Section 2. Aid to Make Good the Damage Caused by Natural Disasters	
	or Exceptional Occurrences	236
	Section 3. Aids Granted to Certain Areas of Germany Affected by	
	the Division of That Country	239
2	Aid Subject to a Discretionary Assessment and an Auticle 107(2) TERM	240
э.	Aid Subject to a Discretionary Assessment under Article 107(3) TFEU	240
	Introduction	240
	Section 1. Regional Aid (Article 107(3)(a) and (c) TFEU)	240
	A. Scope of regional aid 1. Sectoral scope	241241

	Detailed Table of Contents	XV
	2. Material scope: Types of regional aid	244
	2.1. Initial investment and operating aid	244
	2.2. Stricter treatment of large enterprises	246
	3. Geographical scope	247
	3.1. 'a' regions	249
	3.2. 'c' regions	249
	3.3. Regional aid maps	250
	3.4. Maximum aid intensities	250
В.	Architecture of the compatibility rules (GBER, RAG)	252
	1. Delineation between GBER and RAG	252
	2. Architecture of the compatibility assessment	254
	3. General compatibility conditions in the GBER	255
	3.1. Investment aid (Article 14 GBER)	255 256
C	3.2. Operating aid (Article 15 GBER) Application of the common principles and the balancing test under RAG 2014	257
C.	Contribution to a common objective	258
	2. Need for State intervention (market failure vs equity objective)	259
	3. Appropriateness of the aid measure	259
	4. Incentive effect	259
	5. Proportionality	261
	6. Avoidance of undue negative effects	263
	6.1. Manifest negative effects	263
	6.2. Negative effects: Balancing test	264
	7. Transparency of aid	265
	8. Overview of the application of the common principles in	
_	the GBER and RAG	265
	Conclusion	265
	tion 2. Rules on Economic and Financial Crisis (Article 107(3)(b) TFEU)	266
	Introduction	266
В.	The approach of the Commission in the assessment of the compatibility of	267
C	State aid in the financial sector before the crisis of October 2008 The reaction to the 2008 crisis: The temporary framework	207
C.	to support financial institutions and access to finance	268
	The Banking Communication	269
	2. The Recapitalization Communication	271
	3. The Impaired Assets Communication	271
	4. The Restructuring Communication	272
	5. The temporary framework for State aid measures to support access to finance	272
D.	The trade-off between financial stability and competition law in	
	the temporary framework	273
	The modifications of the temporary framework	274
F.	The application of the temporary framework and its scrutiny before	
	the General Court (GC)	275
G.	The New Banking Communication of August 2013: Bridging the gap between	250
	the crisis and a long-term approach	278
	Concluding remarks	279
Sect	ion 3. Research, Development, and Innovation (R&D&I) Aid	200
	(Article 107(3)(c) TFEU)	280
	Introduction	280
В.	Delimitation of the notion of State aid in the R&D&I field	282
	Funding of research organizations and research infrastructures Public funding of non-aconomic activities	282 282
	1.1. Public funding of non-economic activities1.2. Public funding of economic activities	282
	Indirect aid to undertakings through publicly funded research	203
	organizations and infrastructures	284
	organizations and minustractures	201

		3. Research on behalf of undertakings (contract research or research services)	284
		4. Collaboration with undertakings	284
		5. Public procurement of research and development services	285
	C.	Scope	286
		Sectoral scope	286
		2. Material scope: Types of R&D&I aid	286
	D	Architecture of the rules	287
	υ.	Delineation between the GBER and R&D&I Framework	287
			288
		2. General compatibility conditions (GBER)	
		2.1. Exclusions	288
		2.2. Interlinks with EU funds managed by the Commission	200
		(eg Horizon 2020)	289
		2.3. Eligible costs and aid intensities	290
	Ε.	Application of the common principles and the balancing test	291
		Contribution to a common objective	291
		2. Need for State intervention (market failure)	292
		3. Appropriateness of the aid measure	292
		4. Incentive effect	293
		5. Proportionality	294
		6. Avoidance of undue negative effects	295
		6.1. Distortions of the product market	295
		6.2. Location effects	296
		7. Transparency of aid	297
	E	Rescue and restructuring aid	297
		Conclusion	299
		ion 4. The Cultural Derogation (Article 107(3)(d) TFEU)	300
		Introduction	300
	В.	The classification of State aids under Article 107(1) TFEU	201
		and the cultural derogation	301
	C.	The cultural State aid derogation and its (residual) application to	
		the broadcasting sector	303
	D.	The cultural derogation and films	305
	E.	Conclusion	307
4.	Trai	nsport Aid	308
	Α	Different legal bases: Article 93 TFEU (land) vs Articles 106	
		and 107 TFEU (air and maritime)	309
		Land transport	310
		1. Secondary law in land transport: Regulations (EC) Nos 1370/2007	210
		and 1192/1969	310
		2. Guidelines and frameworks	315
		3. Direct application of Article 93 TFEU	316
	C	Air and maritime transport	317
5.	Aid	Exempted from Notification to the Commission:	
	The	General Block Exemption Regulation (GBER)	324
		ion 1. Introduction	324
			324
	Sect	ion 2. Safeguards to Ensure a Positive Balance of	
		the Aid Exempted from Notification	326
		Transparency	326
		Evaluation	327
	C.	Monitoring and withdrawal of the benefit of the GBER	329
		ion 3. Exclusion of Potential Distortive Aid Measures from	
		the Scope of GBER	329
		· · · · · · · · · · · · · · · · · · ·	

Detailed Table of Contents	xvii
Section 4. General Compatibility Criteria under GBER (Chapter I of the G	BER) 330
Section 5. Newly Introduced Categories in the GBER	333
	333
A. Aid to mitigate damages caused by certain natural disasters	
B. Social aid for transport concerning people living in remote regions C. Aid for broadband infrastructures	334
	335
D. Aid for culture, heritage conservation, and audio-visual	335
E. Aid for sport and multifunctional recreational infrastructure	337
F. Local infrastructure	337
Section 6. Conclusion	338
PART IV PROCEDURES BEFORE THE COMMISSION	N
1. The Administrative Procedure—the Privileged Dialogue	341
Introduction	341
Section 1. The Addressee Factor	342
Section 2. Dual Roles	342
	344
Section 3. The 'One Size Fits All' Approach	
Section 4. The Impact of the State Aid Modernization Reform	345
Section 5. The (Lack of) Impact of the Charter of Fundamental Rights	
Conclusion	347
2. The Different Stages in the State Aid Procedure	348
Section 1. Categories of State Aid	348
A. Notified vs non-notified, or unlawful, aid	348
B. New aid vs existing aid	351
Section 2. Initial Stages of the State Aid Inquiry	354
A. Preliminary investigation for notified measures	354
1. Pre-notification: Key features	354
2. Preliminary investigation: Key features	354
3. Preliminary investigation: Role of third parties	355
B. The preliminary examination: Non-notified (unlawful) aid	357
1. Preliminary examination of unlawful aid: Key features	357
2. Preliminary investigation for unlawful aid: Role of third parties	357
2.1. The plus factors and the burden of proof on third parties to	
make a case of breach of their procedural rights	358
2.2. Commission discretion in conducting the investigation as	
a limitation to third-party procedural rights	359
2.3. Conclusion on third-party rights in the preliminary investigation	
phase in case of unlawful aid	360
Section 3. The Legal Status of the Complainant and the Commission's	
Ability to Reject Complaints	361
Section 4. Information Injunctions	364
A. The rules prior to the 2013 Modernization Package	364
B. Information powers after the 2013 Modernization Package	365
Powers to obtain market information	366
2. Sector inquiries	367
3. Conclusion	367
Section 5. Misuse of State Aid	368
A. Parts of the compatibility decision prone to misuse	368
B. Potential for misuse—types of State aid	369
Section 6. The Formal Investigation	371
A. The decision opening a formal investigation	371
B. Access to the file	375
C. Closure of the formal investigation—withdrawal of unlawful aid	377

3.	Rec	overy of Unlawful Aid and the Role of National Courts	381
	Sect	tion 1. Introduction	381
	Sect	tion 2. Legal Framework	382
		tion 3. General Principles of EU Law and their Exceptions	383
		General principles on recovery	383
		1. Compliance with national procedural rules	383
		2. Obligation to take all necessary measures	385
		3. Application of the principle of loyal or sincere cooperation	385
		3.1. Relationships between the Commission and national courts:	206
		The Mediaset case	386
		3.2. Strengthened role of the Commission: The new powers under the SAM initiative	388
	В.	Exceptions to the general principles on recovery	389
	Σ.	Exception 1: General principles of EU law	389
		2. Exception 2: Prescription period for recovery	392
		3. Exception 3: Absolute impossibility to recover the aid	394
	Con	nclusion	396
		PART V JUDICIAL REVIEW BEFORE THE COURT OF	
		JUSTICE OF THE EUROPEAN UNION	
		JUSTICE OF THE EUROPEAN UNION	
1.	Intr	oduction: EU State Aid Litigation in a Quasi-Federal System	401
2.	Dire	ect Actions and Judicial Review before the Union Courts	406
	Sect	tion 1. Introductory Remarks	406
		tion 2. State Aid Measures Amenable to Judicial Review under	100
		Article 263 TFEU	406
	A.	Types of applicants and of challengeable acts	406
		The notion of challengeable act	407
		1. The requirement of producing autonomous binding legal effects from	
		the perspective of Member States/institutions as well as natural	
		and legal persons	407
		2. Challenges of intermediate or preparatory acts	408
		3. Challenges of Commission decisions to initiate	400
		the formal investigation procedure 4. Challenges of Commission letters informing on the rejection of	409
		a State aid complaint	411
	Sect	tion 3. The Requirements of <i>Locus Standi</i>	413
		The concept of (legal) interest in bringing proceedings	413
	В.		417
		1. Preliminary remarks	417
		2. The concept of 'direct concern'	417
		3. The concept of 'individual concern'	419
		3.1. Introductory remark	419
		3.2. Individual concern regarding phase I decisions—the importance	440
		of procedural guarantees granted in phase II	419
	C	3.3. Individual concern regarding phase II decisions Standing to challenge regulatory acts not entailing implementing measures.	424
	C.	Standing to challenge regulatory acts not entailing implementing measures according to Article 263(4) third limb TFEU	428
	D	Admissibility criteria governing the action for failure to act	140
	D.	under Article 265 TFEU	429
	E.	Admissibility criteria governing the action for damages under Articles 268	
		and 340(2) TFEU	431

	Detailed Table of Contents	xix
	Section 4. Scope and Degree of Judicial Review on the Merits,	
	Notably in Annulment Actions	433
	A. Grounds of review and their legally binding nature for the Union judge	433
	B. Comprehensive vs limited scope of judicial review	435
	1. Objective notion of State aid vs complex assessments	435
	2. Judicial review with respect to the <i>Altmark</i> criteria	436
	3. Manifest error: Plausibility check and standard of evidence	437
	4. Judicial review of discretionary assessments governed by 'soft law'	438
	5. Judicial review of conditions/commitments attached to	
	compatibility decisions	439
	C. Distinction between review on the substance and of procedural legality,	
	in particular on the basis of the duty to state reasons	440
	D. Scope and degree of judicial review with regard to (phase I) decisions	
	not to raise objections and opening decisions	445
	E. Judicial review in the context of actions for failure to act under	
	Article 265 TFEU	447
	F. Judicial review in the context of actions for damages under Article 268	
	in combination with Article 340(2) TFEU	448
3.	Decentralized Judicial Review and Enforcement of EU State Aid Rules	450
	Section 1. Union Institutions Involved—'Centralized' Enforcement	450
	Section 2. Decentralized Enforcement—National Courts	451
	Section 3. The Standstill Clause in Article 108(3) Last Sentence TFEU	452
	A. Direct effect	452
	B. Consequences for the national court	453
	C. The development of the competences of the national judge in	
	the course of a State aid procedure	454
	1. Before (or without) notification	454
	2. After initiation of the formal investigation procedure	455
	3. Injunctions	457
	4. Final decisions	457
	D. Consequences of the annulment of an approval of lawful measures	457
	E. Are alternative measures instead of ordering recovery allowed?	458
	F. Procedures to be applied	459
	Section 4. Recovery Decisions	459
	A. Objective of recovery orders	460
	B. National procedural rules apply	460
	C. Impossibility to recover	462
	D. The <i>Deggendorf</i> principle	463
	E. Interest and compound interest	463
	Section 5. Block Exemption Regulations	463
	PART VI GLOBAL APPROACH OF STATE AID LAW—WTO LAW AND EU TRADE DEFENCE INSTRUMENTS	
1.	WTO Subsidy Laws: The International Regulation of State Aid	469
	Section 1. Introduction: Importance of WTO Subsidy Laws Section 2. Brief Outline of WTO Subsidy Laws: Historical Evolution,	469
	Regulatory Framework, Rationales	470
	A. Historical evolution	470
	B. Rationale for controlling subsidies	471
	C. Dispute settlement and remedies	474
	Section 3. The Definition and its Issues: A Matter of Scope	474

	A.	Financial contribution, income/price support, and regulatory measures	476
	В.	Taxation	480
	C.	Provision of goods/services, purchases of goods	482
		What is 'government'?	482
		The benefit	484
	Sect	ion 4. Specificity	487
	Sect	ion 5. Categories of Subsidies	490
	A.	Prohibited subsidies	490
	В.	Actionable subsidies	493
	C.	Non-actionable subsidies	494
	Sect	ion 6. Countervailing Duties: Substance and Procedure	495
	A.	Substantive requirements	495
	В.	Procedural steps	498
	Sect	ion 7. Specific Sectors	500
		Agriculture	500
		Energy	502
	C.	Services	502
	Sect	ion 8. Transparency	503
		ion 9. Conclusion: Lessons and Conflicts	505
2.	Ant	i-subsidy Law and Practice of the European Union	508
	Intro	oduction	508
		ion 1: History and Basics of the EU Anti-subsidy Legislation	509
		History	509
		Overview of the EU Anti-subsidy rules	509
		EU basic Anti-subsidy regulation and the SCM Agreement from	507
	٠.	a comparative perspective	511
	Sect	ion 2: Use of the Anti-subsidy Instrument Since 1995	512
		Overview	512
		Developments and trends	514
		ion 3. Key Elements of a Countervailing Subsidy—the EU Perspective	516
		Financial contribution by a government	516
	11.	Government or public body	517
		2. Forms of financial contribution	520
		2.1. Direct and indirect transfer of funds, revenue foregone,	020
		provision of goods and services	520
		2.2. Government entrustment and direction,	020
		payments via a funding mechanism	521
	В.	Specificity	524
		Subsidies specific to an enterprise or industry or a group	
		of enterprises or industries	524
		2. Regional specificity	527
		3. Export contingent subsidies	528
		3.1. Duty Drawback Schemes	529
		3.2. EOUs and Special Economic Zones (SEZs)	531
		3.3 Import duty exemption	532
		3.4 Export credits	532
		3.5 Tax exemptions	533
		4. Import substitution subsidies	534
	C.	Benefit and subsidy margin calculation	534
		1. Calculation of subsidy amounts in practice	536
		2. Subsidy margin calculation	541
		3. Instances when subsidies were not countervailed	542

Detailed Table of Contents	xxi
Section 4: Injury, Causation, and Union Interest Assessments	543
A. Injury and causation	543
B. Union interest	544
Section 5: Remedies in EU Anti-subsidy Cases, Imposition,	
and Duration of Measures	545
A. Remedies in AS cases	545
B. Imposition of AS duties	546
 Individual and residual duty rates 	546
2. Duty imposition and its modalities	547
C. Reviews	550
Section 6: Procedural Aspects and Judicial Review of EU Anti-subsidy	
Measures	551
A. Procedural aspects	551
B. Judicial review of Anti-subsidy measures	554
Conclusion	555
Annex 1: Correlation Table between the SCM Agreement and	
the EU Basic Anti-subsidy Regulation	556
Annex 2: EU Anti-subsidy Cases Since 1995	559
Index	565

Table of Cases

n =footnote.

COURT OF JUSTICE OF THE EUROPEAN UNION (CJEU) – COURT OF JUSTICE AND GENERAL COURT

Joined Cases C-7/56 and C-3 to 7/57 Algera and others v Common Assembly [1957/58] ECR 39 \ldots .41, 54 n
C-9/56 Meroni v ECSC High Authority [1958] ECR English special edition, 13357n
Joined Cases C-1/57 and C-14/57 Société des usines à tubes de la Sarre [1957] ERT 105
C-30/59 Steenkolenmijnen v High Authority 1961 ECR 1
Joined Cases C-42/59 and C-49/59 SNUPAT v High Authority [1961] ECR 5354n
C-25/62 Plaumann v Commission [1963] ECR 95
C-32/62 Alvis v Council [1963] ECR 49
Joined Cases C-56/64 and C-58/64 Consten and Grundig v Commission [1966] ECR 299
Joined Cases C-6/69 and C-11/69 Commission v French Republic [1969] ECR 523
C-47/69 France v Commission ('Institut textile de France') ECLI:EU:C:1970:60
C-22/70 Commission v Council (EATR case) [1971] ECR 263
Joined Cases C-21/72 to C-24/72 International Fruit Company NV v Produktschap voor Groenten en Fruit [1972] ECR 121937, 37n
C-70/72 Commission v Federal Republic of Germany [1973] ECR 813 (Kohlgesetz case) 21–2, 382, 460
C-120/73 Gebrüder Lorenz GmbH v Germany [1973] ECR 1471 21–2, 29, 43, 349, 420n, 452
C-121/73 Markmann KG v Federal Republic of Germany and Land Schleswig-Holstein [1973] ECR 1495
C-122/73 Nordsee, Deutsche Hochseefischerei GmbH v Federal Republic of Germany and Land Rheinland-Pfalz [1973] ECR 1511
C-127/73 BRT v SABAM [1974] ECR 313
C-141/73 Fritz Lohrey v Federal Republic of Germany and the Land Hessen [1973] ECR 152722n
C-148/73 Louwage v Commission [1974] ECR 8160
C-173/73 Italy v Commission [1974] ECR 70984, 136–7n, 154n, 166, 170, 175n, 176n
C-181/73 R & V Haegeman v Belgian State [1974] ECR 44937n
C-105/75 Giuffrida v Council [1976] ECR 1395
C-33/76 Rewe-Zentralfinanz eG and Rewe-Zentral AG v Landwirtschaftskammer für das Saarland [1976] ECR 1989
C-45/76 Comet BV v Produktschap voor Siergewassen [1976] ECR 2043402n
C-74/76 Ianelli & Volpi Spa v Ditta Paola Meroni [1977] 2 CMLR 688
C-78/76 Steinike und Weinlig v Germany [1977] ECR 595 67n, 89n, 154, 166, 207n, 454, 517
C-85/76 Hoffmann-La Roche v Commission [1979] ECR 46147
C-82/77 Public Prosecutor of the Netherlands v Van Tiggele [1978] ECR 25
C-106/77 Amministrazione delle Finanze dello Stato v Simmenthal SpA [1978] ECR 629
C-156/77 Commission v Belgium [1978] ECR 1881
C-98/78 Racke v Hauptzollamt Mainz [1979] ECR 6960n
C-130/78 Salumificio di Cornuda SpA v Amministrazione delle Finanze dello Stato [1979] ECR 867 461n
C-243/78 Simmenthal v Commission [1980] ECR 593
Joined Cases C-33/79 and C-75/79 Kuhner v Commission [1980] ECR 1677
C-61/79 Amministrazione delle finanze dello Stato v Denkavit Italiana Srl. [1980] ECR 1205
C-73/79 Commission v Italy (sugar levy case) ECLI:EU:C:1980:129
C-138/79 Roquette v Council [1980] ECR 3333
C-730/79 Philip Morris v Commission [1980] ECR 2671

C-125/80 Arning v Commission [1981] ECR 2539
$Joined\ Cases\ C-188\ to\ C-190/80\ France,\ Italy\ and\ United\ Kingdom\ v\ Commission\ [1992]\ ECR\ 2545.\ldots.\ 110nc$
C-279/80 P Commission v Netherlands [2011] ECR I-551
C-60/81 IBM v Commission [1981] ECR 2639
C-76/81 SA Transporoute et Travaux v Minister of Public Works [1982] ECR 457
C-96/81 Commission v Netherlands [1982] ECR 1791
C-104/81 Hauptzollamt Mainz v Kupferberg [1982] ECR 364137n
C-210/81 Demo-Studio Schmidt v Commission [1983] ECR 3045
$ Joined \ Cases \ C-213 \ to \ C-215/81 \ Nord deutsches \ Vieh-und \ Fleischkontor \ [1982] \ ECR \ 3583 \ldots 65, 68, 206, 210nn \ Annual Market \ A$
C-244/81 Commission v Ireland [1982] ECR 4005
C-11/82 Piraiki-Patraiki and Others v Commission [1985] ECR 207
C-64/82 Tradax v Commission [1984] ECR 1359
C-84/82 Federal Republic of Germany v Commission [1984] ECR 01451
C-191/82 Fediol v Commission [1983] ECR 2913
$Joined \ Cases \ C-205/82 \ to \ C-215/82, Deutsche \ Milchkontor \ and \ Others \ v \ Germany \ [1983] \ ECR \ 02633 \ \dots 460n \ (1983) \ CR \ 02633 \ \dots 10000000000000000000000000000000000$
Joined Cases C-296 and C-318/82 Netherlands and Leeuwarder Papierfabrie v Commission [1985] ECR 809
C-323/82 Intermills v Commission [1984]ECR 3809105n, 106n, 108-9, 112, 166, 176n, 421, 421n
C-14/83 von Colson and Kamann v Land Nordrhein-Westfalen [1984] ECR 1891
Joined Cases C-91/83 and C-127/83 Heineken Brouwerijen v Inspecteur der Vennootschapsbelasting, Amsterdam and Utrecht [1984] ECR I-3435
C-145/83 Stanley Adams v Commission [1985] ECR 3539
C-240/83 Procureur de la République v Association de défense des brûleurs d'huiles usagées (ADBHU) [1985] ECR 531
C-290/83 <i>Commission v France</i> (subsidies for poor farmers) [1985] ECR 439
C-294/83 Les Verts v European Parliament [1986] ECR 1339
C-18/84 Commission v France [1985] ECR 1339
C-42/84 Remia BV and Verenigde Bedrijven Nutricia v Commission [1985] ECR 2545
C-52/84 Commission v Belgium [1986] ECR 89
C-103/84 Commission v Italy [1985] ECR 1759
C-107/84 Commission v Germany [1985] ECR 2655
Joined Cases C-142/84 and C-156/84 BAT and Reynolds v Commission [1987] ECR 448745n, 57n
C-169/84 Cofaz and Others v Commission [1986] ECR 391
Joined Cases C-209 to C-213/84 Nouvelles Frontières (Criminal proceedings against Lucas
Asjes and others) [1986] ECR1457
C-222/84 Johnston v Chief Constable of the Royal Ulster Constabulary [1986] ECR 1651
C-234/84 Belgium v Commission (Meura case) [1986] ECR 2263 43n, 108, 109–12, 113, 122, 173n, 174n
C-248/84 Germany v Commission [1986] ECR 2263
C-24/85 Spijkers v Gebroders Benedik Abbatoir CV [1986] ECR 1123
C-40/85 Belgium v Commission (Boch II case) [1986] ECR 2340
Joined Cases C-67, 68 and 70/85 Van der Kooy v Commission [1988] ECR 219 68n, 72, 166n, 174n, 426n
C-118/85 Commission v Italy [1987] ECR 2599
C-223/85 RSV v Commission [1987] ECR 4617
C-259/85 France v Commission [1987] ECR 4393
C-310/85 Deufil v Commission [1987] ECR 901
C-12/86 Demirel v Stadt Schwabisch Gmünd [1987] ECR 3719
C-57/86 <i>Greece v Commission</i> (interest on export credits) [1988] ECR 2855
C-66/86 Ahmed Saeed Flugreisen v Commission [1989] ECR 803
C-240/86 Commission v Greece [1988] ECR 1835
C-263/86 Belgian State v René Humbel and Marie-Thérèse Edel [1988] ECR 536580
Case C-324/86 Tellerup [1998] FCR 739 I-3547 185n

C-31/87 Gebroeders Beenjes BV v The Netherlands [1989] ECR 4365
Joined Cases C-62/87 and C-72/87 Walloon Government and Glaverbel SA v Commission [1988] ECR 1573
C-63/87 Commission v Greece [1988] ECR 2875
C-70/87 Fediol v Commission (Fediol III) [1989] ECR 1781
C-94/87 Commission v Germany [1989] ECR 175
C-102/87 France v Commission (loan grant to Fonds industriel de modernisation) [1988] ECR 4067 154-5
C-142/87 Belgium v Commission (Tubemeuse) [1990] ECR I-959 64n, 112, 153n, 155n, 159, 171, 173n, 174n, 176n, 383n, 458n, 460, 491
C-301/87 France v Commission (Boussac Saint Frères) [1990] ECR I-307
C-10/88 Italy v Commission [1990] ECR I-1229
C-68/88 Commission v Greece [1989] ECR 2965
C-103/88 Fratelli Costanzo S.p.A. v Comune di Milano [1989] ECR 1839
C-152/88 Sofrimport v Commission [1990] ECR I-2477
C-202/88 France v Commission [1991] ECR I-1223
C-303/88 Italy v Commission (ENI-Lanerossi) [1991] ECR I-1433
C-350/88 Delacre and Others v Commission [1990] ECR I-395
C-5/89 Commission v Germany [1990] ECR I-3437
T-61/89 Dansk Pelsdyravlerforening v Commission [1992] ECR II-1931
C-70/89 Nakajima v Council [1991] ECR I-2069
C-213/89 R, ex p Factortame and Others, v Secretary of State for Transport [1990] ECR I-2433
C-261/89 Italy v Commission [1991] ECR I-4437
C-296/89 Impresa Dona Alfonso di Dona Alfonso & Figli s.n.c. v Consorzio per lo Sviluppo
Industriale del Comune di Monfalcone, Judgment of 18 June 1991
C-305/89 Italy v Commission (Alfa Romeo) [1991] ECR I-1603
C-360/89 Commission v Italy [1992] ECR 3401
C-16/90 Nölle v Hauptzollamt Bremen-Freihafen [1991] ECR I-516360n
C-24/90 Automec v Commission [1992] ECR II-2223
C-41/90 Klaus Höfner and Fritz Elser v Macrotron GmbH [1991] ECR I-197975, 75n
T-46/90 Devillez and Others v European Parliament [1993] ECR II-699
Joined Cases C-48/90 and C-66/90 Netherlands and Others v Commission [1992] ECR I-565 46, 346n
Joined Cases C-87/90 to C-89/90 Verholen and Others v Sociale Verzekeringsbank Amsterdam [1991] ECR I-3757
C-208/90 Emmott v Minister for Social Welfare and Attorney General [1991] ECR I–4269
C-255/90 P Burban [1992] ECR I-2253
C-269/90 Technische Universität München v Hauptzollamt München-Mitte [1991] ECR I-5469
Joined Cases C-271/90, C-281/90, and C-289/90 Spain, Belgium and Italy v Commission [1992]
ECR I-5833
C-312/90 Spain v Commission [1992] ECR I-4117. 409n, 456
C-313/90 CIRFS and Others v Commission [1992] ECR I-4117
Joined Cases C-324/90 and C-342/90 Germany and Pleuger Worthington v Commission [1994]
ECR I-1173
national des négociants et transformateurs de saumon v France [1991] ECR I-5505
452 <i>n</i> , 453, 458
C-364/90 Italy v Commission [1993] ECR I-2097
C-17/91 Lornov and Others v Belgium [1992] FCR I-6523 452n

C-29/91 Dr Sophie Redmond Stichting v Bartol [1992] ECR 3189
C-47/91 Italy v Commission (Italgrani) [1994] ECR I-4635
Joined Cases C-72/91 and 73/91 Firma Sloman Neptun Schiffahrts AG v Seebetriebsrat Bodo Ziesemer der Sloman Neptun Schiffahrts AG [1993] ECR I-887
C-97/91 Borelli SpA v Commission [1992] ECR I-631
Joined Cases C-159/91 and C-160/91 Christian Poucet v Assurances Générales de France and Caisse Mutuelle Régionale du Languedoc-Roussillon [1993] ECR I-637
C-183/91 Commission v Greece [1993] ECR I-3131
C-189/91 Kirsammer-Hack v Nurhan Sidal [1993] ECR I-6185
C-198/91 William Cook plc v Commission of the European Communities [1993] ECR I-2487 29n, 419, 420, 420n, 421–2
C-209/91 Rask v ISS Kantinservice [1993] ECR 5735
$C-225/91\ \textit{Matra v Commission}\ ECLI: EU: C:1993:239.\dots \\ 230, 258n, 288-9, 291, 419, 420, 420n, 421-230, 258n, 288-9, 291, 419, 420, 420n, 421-230, 420n, 420n,$
C-271/91 Marshall v Southampton and South West Hampshire Area Health Authority [1993] ECR I-4367403n
C-316/91 Parliament v Council [1994] ECR I-625
C-320/91 Criminal proceedings against Paul Corbeau [1993] ECR I-253389n, 92
C-325/91 France v Commission [1993] ECR I-3283
C-6/92 Federmineraria and Others v Commission [1993] ECR I-6357
T-7/92 Asia Motor France and Others v Commission [1993] ECR II-669
T-65/92 Arauxo-Dumay v Commission [1993] ECR II-597
Joined Cases T-98/92 and T-99/92 Di Marzio and Lebedef v Commission [1994] ECR II-541
C-188/92 TWD Textilwerke Deggendorf [1994] ECR I-833
Joined Cases C-278/92 to C-280/92 Spain v Commission [1994] ECR 4103 113n, 114n, 120, 122, 123,
152, 166, 173 <i>n</i> , 174 <i>n</i> , 299, 381, 383 <i>n</i>
C-364/92 SAT Fluggesellschaft mbH v Eurocontrol [1994] ECR I-43
C-382/92 Commission v United Kingdom [1994] ECR 2435
C-387/92 Banco Exterior de España SA v Ayuntamiento de Valencia [1994] ECR I-87 68, 69n, 85, 89, 172n, 174n
C-392/92 Schmidt v Spar und Leihkasse der fruherer Amter Bordersholm, Kiel und Cronshagen [1994] ECR 1320
C-7/93 Bestuur van het Algemeen Burgerlijk Pensioenfonds v GA Beune [1994] ECR I-04471172n
C-39/93 P SFEI and Others v Commission [1994] ECR I-2681
C-42/93 Spain v Commission [1994] ECR I-4175
C-44/93 <i>Namur-Les Assurances du Crédit SA</i> v Office National du Ducroire and the Belgian State [1994] I-03829
C-56/93 Belgium v Commission [1996] ECR I-723
C-135/93 Spain v Commission [1995] ECR I-1651
Joined Cases T-244/93 and T-486/93 Textilwerke Deggendorf v Commission (TWD) 1995 ECR II-2265
C-280/93 Germany v Council [1994] ECR I-4973
C-312/93 Peterbroeck, Van Campenhout & Cie SCS v Belgium [1995] ECR I-4599402n
C-324/93 R v The Secretary of State for the Home Department, ex parte Evans Medical Ltd and Macfarlan Smith Ltd [1995] ECR 563
Joined Cases C-329/93, C-62/95, and C-63/95, Bremer Vulkan and Others v Commission [1996] ECR I-5151
C-348/93 Commission v Italy [1995] ECR I-673
C-350/93 Commission v Italy [1995] ECR I-699
C-392/93 R v HM Treasury, ex parte British Telecommunications PLC, OJ 1993, C 287/6
C-431/93 van Schijndel and Others v Stichting Pensioenfonds voor Fysiotherapeuten [1995] ECR I-4705
T-435/93 ASPEC and Others v Commission [1995] ECR II-1281
T.442/93 A AC and Others v Commission [1995] FCR II-1329 418n

Joined Cases T-447 to T-449/93 Associazione Italiana Tecnico Economica del Cemento (AITEC	
v Commission [1995] ECR II-1971.	
T-450/93 Lisrestal v Commission [1994] ECR II-1177	
T-459/93 Siemens v Commission [1995] ECR II-1675	153-4,225n
C-39/94 Syndicat français de l'Express international (SFEI) et al v La Poste et al [1996] ECR I-3547	51 <i>n</i> , 452, 453
C-48/94 Rygaard v Stro Molle Akustik [1995] ECR 2745	185 <i>n</i>
C-61/94 Commission v Germany [1996] ECR I-3989	37n
T-67/94 Ladbroke Racing v Commission [1998] ECR II-1	
T-95/94 Sytraval and Brink's France v Commission [1995] ECR II-2651	
C-122/94 Commission v Council [1996] ECR I-881	
T-157/94 Commission v the Netherlands [1997] ECR I-5699	78n, 229-30
T-158/94 Commission v Italy [1997] ECR I-5789	91, 178 <i>n</i>
T-159/94 Commission v France [1997] ECR I-5815	91, 178 <i>n</i>
T-160/94 Commission v Spain [1997] ECR I-5851	
T-167/94 Nölle v Council and Commission [1995] ECR II-2589	
C-241/94 France v Commission [1996] ECR I-4551	
C-244/94 Fédération Française des Sociétés d'Assurance, Société Paternelle-Vie et al. v Ministè de l'Agriculture et de la Pêche [1995] ECR I-4013	re
T-260/94 Air Inter v Commission [1997] ECR II-997	
C-311/94 IJssel-Vliet [1996] ECR I-5023	
T-358/94 Air France v Commission [1996] ECR II-2109 56, 65, 66, 67n, 70, 71n, 1	
Joined Cases T-371/94 and T-394/94 British Airways and Others v Commission of the Europea Communities [1998] ECR II-2405	n
T-380/94 AIUFFASS et AKT v Commission of the European Communities [1996] ECR II-2169	26n, 51n,
Therefore W. L. G. L. L. L. L. G. L.	425n, 437n
T-398/94 Kahn Scheppvaart BV v Commission of the European Communities [1996] ECR II-47	
T-11/95 BP Chemicals v Commission [1998] ECR II-3235	
C-24/95 Land Rheinland-Pfalz v Alcan Deutschland [1997] ECR I-1591	
T-24/95 Jacobsen v Council and Commission [1997] ECR I-1591	
C-32/95 P Commission v Lisrestal [1996] ECR I-5373	
C-52/95 Commission v France [1995] ECR I-4443	
C-57/95 France v Commission [1997] ECR I-01627.	
T-81/95 Interhotel v Commission [1997] ECR II-1265.	
T-106/95 FFSA v Commission [1997] ECR II-0229	
C-120/95 Nicolas Decker v Caisse de maladie des employés privés [1998] ECR I-1831	
Joined Cases T-129/95, T-2/96, and T-97/96 Neue Maxhütte Stahlwerke and Others v Commiss [1999] ECR II-17	
T-149/95 Ducros v Commission [1997] ECR II-2031	418 <i>n</i>
C-166/95 P Commission v Daffix [1997] ECR I-983	441–2 n
C-169/95 Spain v Commission [1997] ECR I-135	2, 248 <i>n</i> , 461
C-178/95 Wiljo NV v Belgium [1997] ECR I-585	405n
$T-214/95\ V laamse\ Gewest\ v\ Commission\ [1998]\ ECR\ II-71726n, 51n, 153-4, 11$	53n, 443-4n
C-265/95 Commission v France [1997] ECR I-6959	462
C-280/95 Commission v Italy [1998] ECR I-259.	395, 462
C-282/95 P Guérinautomobiles v Commission [1997] ECR I-1503	43n
C-292/95 Spain v Commission [1997] ECR I-1931	227
C-343/95 Diego Calì & Figli Srl v Servizi ecologici porto di Genova SpA (SEPG) [1997] ECR I-15	
C-355/95 P Textilwerke Deggendorf GmbH (TWD) v Commission [1997] ECR I-2549	.56, 462, 463
C-367/95 P Commission v Chambre syndicale nationale des entreprises de transport de fonds e valeurs (Sytraval) and Brink's France SARL [1998] ECR I-1719 29, 29n, 43, 59–60, 6	
421n 433n 434 435n 441-2n	

T-14/96 Bretagne Anglaterre Irlande (BAI) v Commission [1999] ECR II-141
$T-16/96\ City flyer\ Express\ Ltd\ v\ Commission\ [1998]\ ECR\ II-757\$
T-17/96 TF1 v Commission [1999] ECR II-1757
T-25/96 Arbeitsgemeinschaft Deutscher Luftfahrt-Unternehmen and Hapag-Lloyd v Commission [1997] ECR II-363
C-35/96 Commission v Italy [1998] ECR I-385175n
C-44/96 Mannesmann Anlangenbau Austria AG et al. v Strohal Rotationsdurck GesmbH [1998] ECR 73
T-65/96 Kish Glass [2000] ECR II-1885
C-67/96 Albany International BV v Stichting Bedrijfspensioenfonds Textielindustrie [1999] ECR I-5751
T-95/96 Gestevisión Telecinco v Commission [1998] ECR II-3407
T-102/96 Gencor v Commission [1999] ECR II-753
Joined Cases T-126/96 and T-127/96 Breda Fucine Meridionali and Others v Commission [1998] ECR II-3437
Joined Cases T-132/96 and T-143/96 Freistaat Sachsen and Others v Commission [1999] ECR II-3663
C-149/96 Portugal v Council [1999] ECR I-8395
C-158/96 Raymond Kohll v Union des caisses de maladie [1998] ECR I-1931
C-266/96 Corsica Ferries France v Gruppo Antichi Ormeggiatori del porto di Genova and Others [1998] ECR I-3949
C-288/96 Germany v Commission [2000] ECR I-8237
C-301/96 Germany v Commission [2003] ECR I-9919
C-342/96 Spain v Commission [1999] ECR I-2459
C-343/96 Dilexport Srl v Amministrazione delle Finanze dello Stato [1999] ECR I-579
C-360/96 Gemeente Arnhem Gemeente Rheden v BFI Holding BV [1998] ECR 6821 161n, 170, 174, 182n
C-361/96 Société générale des grandes sources d'eaux minerals françaises v Bundesamt für Finanzen [1998] ECR I-3495
C-386/96 P Dreyfus v Commission [1998] ECR I-2309
C-404/96 P Glencore Grain v Commission [1998] ECR I-2435
C-412/96 Kainuun Liikenne Oy [1998] ECR 1515171n
C-415/96 Spain v Commission [1998] ECR I-6993
T-37/97 Forges de Clabecq v Commission [1999] ECR II-859
T-46/97 SIC v Commission [2000] ECR II-2125
Joined Cases C-52/97, C-53/97, and C-54/97 <i>Epifania Viscido and Others v Ente Poste Italiane</i> [1998] ECR I-2629
C-75/97 Belgium v Commission [1999] ECR I-03671
T-110/97 Kneissl Dachstein Sportartikel AG v Commission [1999] II-2881
Joined Cases C-115/97 to C-117/97 Brentjens' Handelsonderneming BV v Stichting Bedrijfspensioenfonds voor de Handel in Bouwmaterialen [1999] ECR I-6025
C-174/97 P FFSA v Commission [1998] ECR I-1303
Joined Cases T-186/97, T-187/97, T-190/97 to T-192/97, T-210/97, T-211/97, T-216/97, T-217/97, T-218/97, T-279/97, T-280/97, T-293/97 and T-147/99 Kaufring e.a. v Commission [2001] ECR II-1337
T-189/97 Comité d'entreprise de la Société française de production and Others v Commission [1998] ECR II-335
Joined Cases T-197/97 and T-198/97 Weyl Beef Products a.o. v Commission EU:T:2001:28 Kaufring e.a. v Commission [2001] ECR II-1337
C-200/97 Ecotrade v Altiforni e Ferriere di Servola [1998] ECR I-7907
C-204/97 Portugal v Commission EU:C:2001:233230
Joined Cases T-204/97 and T-270/97 EPAC v Commission [2000] ECR II-2267
T-213/97 Eurocoton v Council [2003] ECR I-10091

T-217/97 Maatschappij Drijvende Bokken BV v Stichting Pensioenfonds voor de Vervoer- en
Havenbedrijven [1999] ECR I-6121,
C-224/97 Ciola v Land Vorarlberg [1999] ECR I-2530
C-256/97 Déménagements-Manutention Transport SA (DMT) [1999] ECR I-03913 131, 154n, 174n
C-265/97 P VBA v Florimex and Others [2000] ECR I-2061
T-288/97 Regione Friuli Venezia Giulia v Commission [2001] ECR II-1169
C-295/97 Piaggio [1999] ECR I-3735
T-296/97 Alitalia v Commission [2000] ECR II-3871
Joined Cases T-298/97, 312-313/97, 315/97, 600-607/97, 1/98, 3-6/98, and 23/98 Alzetta Mauro v Commission [2000] ECR II-2319
C-372/97 Italy v Commission [2001] ECR I-3679
C-404/97 Commission v Portugal (EPAC aid) [2000] ECR I-4897
T-613/97 UFEX and Others v Commission [2000] ECR II-4055
T-9/98 Mitteldeutsche Erdöl-Raffinerie v Commission [2001] ECR II-3367
Joined Cases C-15/98 and 105/99 Italian Republic and Sardegna Lines v Commission [2000] ECR I-8855
T-73/98 Prayon-Rupel v Commission [2001] ECR II-867
C-83/98 P France v Ladbroke Racing and Commission [2000] ECR I-3271 56, 57n, 67n, 69, 72, 207–8
C-99/98 Austria v Commission [2001] I-1142
C-106/98 P Comité d'entreprise de la Société française de production and Others v Commission [2000] ECR I-3659
C-107/98 Teckal Srl v Comune di Viano and Azienda Gas-Acqua Consorziale (AGAC) di Reggio Emilia[1999] ECR I-8121
T-127/98 UPS Europe v Commission [1999] ECR II-2633
C-128/98 Aéroports de Paris v Commission [2000] ECR II-3929
C-156/98 Germany v Commission [2000]ECR I-6857
T-178/98 Fresh Marine v Commission [2000] ECR II-3331
Joined Cases C-180/98 to C-184/98 Pavel Pavlov et al v Stichting Pensioenfonds Medische Specialisten [2000] ECR I-6451
C-206/98 Belgium v Commission [2000] ECR I-3509
C-324/98 Telaustria Verlags GmbH and Telefonadress GmbH v Telekom Austria AG [2000] ECR I-10745
C-351/98 Spain v Commission [2002] ECR I-8031 38, 51n, 53-4, 132n, 152-3, 153n, 154, 155n, 159nn
C-352/98 P Bergaderm and Goupil v Commission [2000] ECR I-5291
C-379/98 PreussenElektra AG v Schhleswag AG [2001] ECR I-2099
486–7,517
C-380/98 R v HM Treasury, ex parte University of Cambridge [2000] ECR 8035
C-390/98 HJ Banks & Co. Ltd v The Coal Authority and Secretary of State for Trade and Industry
[2001] ECR I-6117
C-462/98 P Mediocurso v Commission [2000] ECR I-7183
T-35/99 Keller and Keller Meccanica v Commission of the European Communities [2002] ECR II-261
T-36/99 Lenzing v Commission [2004] ECR II-359759
T-54/99 max.mobil v Commission [2002] ECR II-313
T-55/99 CETM v Commission, judgment of 29 September 2000
C-58/99 Mukand and Others v Council [2001] ECR II-2521
Joined Cases T-83/99, T-84/99, and T-85/99 <i>Ripa di Meana v European Parliament</i> [2000] ECR II-3493
C-94/99 ARGE Gewässerschutzt [2000] ECRI-11037
T-126/99 Graphischer Maschinenbau v Commission [2002] ECR II-2427
Joined Cases T-127/99, T-129/99, and T-148/99 Daewoo Electronics Manufacturing España SA (Demesa)
and Tarritoria Histórica de Álaya Dinutación Faral de Álaya y Commission (2002) FCD II 1775 145n

C-143/99 Adria-Wien Pipeline GmbH and Wietersdorfer & Peggauer Zementwerke GmbH v
Finanzlandesdirektion für Kärnten [2001] ECR I-08365 133n, 135-6, 135n, 136-7n, 140,
140 <i>n</i> , 148 <i>n</i> , 150, 481
C-157/99 B.S.M. Geraets-Smits v Stichting Ziekenfonds VGZ and H.T.M. Peerbooms v Stichting
CZ Groep Zorgverzekeringen [2001] ECR I-547380n, 132n
T-187/99 Agrana Zucker und Stärke AG v Commission [2001] ECR II-1587
T-192/99 Thyssen Stahl v Commission [2003] ECR I-10821
T-196/99 Area Cova and Others v Council and Commission [2001] ECR II-359741n
T-197/99 Gooch v Commission [2000] ECR-SC I-A-271 and II-1247
C-205/99 Asociación Profesional de Empresas Navieras de Líneas Regulares (Analir) et al. v Administración General del Estado [2001] II-127190–1n, 98n, 320
$C-206/99\ \textit{M\'etropole T\'el\'evision v Commission}\ [2001]\ ECR\ II-2707\\ 43n, 60, 390n$
$C-223/99\ Agora\ Srl\ v\ Ente\ Autonomo\ Fiera\ Internazionale\ di\ Milano\ [2001]\ ECR\ 3605.\ldots\ 161n, 170, 174n, 17$
Joined Cases T-228/99 and T-233/99 Westdeutsche Landesbank Girozentrale and Land Nordrhein-Westfalen v Commission [2003] ECR II-435 41n, 66n, 122, 342, 441n
C-234/99 Niels Nygård v Svineafgiftsfonden and others [2002] ECR I-3657
C-237/99 Commission v France [2001] ECR 934
Joined Cases C-238/99 P, C-244/99 P, C-245/99 P, C-247/99 P, C-250/99 P to C-252/99 P, and
C-254/99 P Limburgse Vinyl Maatschappij and Others v Commission [2002] ECR I-8375
C-239/99 Nachi Europe GmbH v Hauptzollamt Krefeld [2001] ECR I-1197
C-260/99 Excelsior Snc di Pedrotti Runa & C v Ente Autonomo Fiera Internazionale di Milano [2001] ECR 3605
C-261/99 Commission v France [2001] ECR I-2537395n
Joined Cases T-269/99, T-271/ 99, and T-272/99 Diputación Foral de Guipúzcoa v Commission [2002] ECR II-4217
C-270/99 P Z v European Parliament [2001] ECR I-9197
C-307/99 OGT v Hauptzollamt St. Annen [2001] ECR I-3159
C-310/99 Italy v Commission (employment aid) [2002] ECR I-2289
Joined Cases C-328/99 and C-399/00 Italy and SIM 2 Multimedia SpA v Commission [2003] ECR I-4035
C-334/99 Germany v Commission [2003] ECR I-1139
C-340/99 Arne Mathisen AS v Council [2002] ECR II-2905
C-342/99 Airtours v Commission [2002] ECR II-2585
Joined Cases T-346/99 to T-348/99 Diputación Foral de Álava and Others v Commission [2002] ECR II-4259
T-354/99 Kuwait Petroleum v Commission [2006] ECR II-1475
C-382/99 Netherlands v Commission [2002] ECR I-5163
C-400/99 Italy v Commission (Tirrenia) [2001] ECR I-7303
C-470/99 Universale-Bau and Others v Entsorgungsbetriebe Simmering GesmbH [2002]
ECR I-11617
C-475/99 Firma Ambulanz Glöckner v Landkreis Südwestpfalz [2001] ECR I-8089
C-482/99 France v Commission (Stardust Marine) [2002] ECR I-4397 67, 68n, 70, 71–2, 72n,
120, 205, 208n C-499/99 Commission v Spain [2002] ECR I-6031
C-513/99 Concordia Bus Filandia Oy Ab v Helsingin Kaupunki et HKL-Bussiliikenne [2002] ECR 213 185n
C-23/00 P Council v Boehringer [2002] ECR I-1873
Joined Cases C-27/00 and C-122/00 Omega Air and Others [2002] ECR I-2569
C-50/00 P Unión de Pequeños Agricultores v Council [2002] ECR I-2309
C-53/00 Ferring SA v Agence centrale des organismes de sécurité sociale (ACOSS) [2001]
ECR I-9067
Joined Cases C-57/00 P and C-61/00 P Freistaat Sachsen, Volkswagen AG and Volkswagen Sachsen GmbH v Commission [2003] ECR I-09975

$C-66/00\ Commission\ v\ Italy\ (banking\ tax\ advantages)\ [2005]\ ECR\ I-10901, ECR\ I-9067\155n$
Joined Cases C-74/00 P and C-75/00 P Falck and Acciairie di Bolzano v Commission [2002] ECR I-7869
C-76/00 P Petrotub and Republica v Council [2003] ECR I-79
Joined Cases T-92/00 and T-103/00 Diputación Foral de Álava v Commission [2002] ECR I-5553; Componeta Oyj [2011] OJ L230/69
C-113/00 <i>Spain v Commission</i> (agricultural product rebate) [2002] ECR I-7601 154 <i>n</i> , 159 <i>n</i> , 373
C-114/00 Spain v Commission [2002] ECR I-7657
T-190/00 Regione Siciliana v Commission, judgment of 27 November 2003
C-209/00 Commission v Germany [2002] ECR I-11695
C-214/00 Commission v Spain [2003] ECR I-4667
C-218/00 Cisal di Battistello Venanzio & C. Sas v Istituto nazionale per l'assicurazione contro gli infortuni sul lavoro (INAIL) [2002] ECR I-691
T-218/00 Cooperative Mare Azzurro and Others v Commission (CJEU, 22 January 2013)
Joined Cases T-226/00 and T-227/00 Far Eastern Textiles Ltd and Nan Ya Plastics Corporation v Council (terminated)
C-242/00 Germany v Commission [2002] ECR I-5603
Joined Cases T-254/00, T-270/00 and T-277/00 <i>Hotel Cipriani e.a. v Commission</i> [2008] ECR II-3269
C-277/00 Germany v Commission [2004] ECR I-3925
C-278/00 <i>Greece v Commission</i> (agricultural cooperative debts) [2004] ECR I-3997 67, 159 <i>n</i> , 234, 237–8
C-279/00 Commission v Italy [2002] ECR I-1425
C-280/00 Altmark Trans GmbH and Regierungspräsidium Magdeburg v Nahverkehrsgesellschaft
Altmark GmbH [2003] ECR I-7747
154–5, 155 <i>n</i> , 159, 172 <i>n</i> , 173, 178, 225 <i>n</i> , 308, 314, 316, 322, 323, 436–7, 440, 446–7, 487
C-283/00 Commission v Spain [2003] ECR I-11697
C-298/00 Italy v Commission (road transport aid) [2004] ECR I-4087
$C-308/00\ Commission\ v\ Salzgitter\ [2004]\ ECR\ II-01933$
T-310/00 MCI v Commission [2004] ECR II-3253
C-312/00 P Commission v Camar and Tico [2002] ECR I-11355
T-318/00 Freistaat Thüringen v Commission [2005] ECR II-4179,
C-323/00 P DSG Dradenauer Stahlgesellschaft v Commission [2002] ECR I-3919,343n
6 323/001 Dod Drawenauer staningesensensyt v Germinssion [2002] EGRI 3717,
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877
· ·
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n T-369/00 Département du Loiret (France) v Commission [2007] ECR II-851 .393n C-373/00 Adolf Truley GmbH v Bestattung Wien GmbH [2003] ECR-193 .164n C-409/00 Spain v Commission, judgment of 13 February 2003 .38, 140n, 159n
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n T-369/00 Département du Loiret (France) v Commission [2007] ECR II-851 .393n C-373/00 Adolf Truley GmbH v Bestattung Wien GmbH [2003] ECR-193 .164n C-409/00 Spain v Commission, judgment of 13 February 2003 .38, 140n, 159n C-457/00 Belgium v Commission (Verlipack) [2003] ECR I-6931 .52, 120-1
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
$ \begin{array}{llllllllllllllllllllllllllllllllllll$
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n T-369/00 Département du Loiret (France) v Commission [2007] ECR II-851 .393n C-373/00 Adolf Truley GmbH v Bestattung Wien GmbH [2003] ECR-193 .164n C-409/00 Spain v Commission, judgment of 13 February 2003 .38, 140n, 159n C-457/00 Belgium v Commission (Verlipack) [2003] ECR I-6931 .52, 120-1 C-5/01 Belgium v Commission [2002] ECR I-3452 .175n, 176n C-13/01 Safalero Srl v Prefetto di Genova [2003] ECR I-8679 .459n C-18/01 Arkkitehtuuritoimisto Riitta Korhonen Oy, Arkkitehtitoimisto Pentti Toivanen Oy,
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n T-369/00 Département du Loiret (France) v Commission [2007] ECR II-851 .393n C-373/00 Adolf Truley GmbH v Bestattung Wien GmbH [2003] ECR-193 .164n C-409/00 Spain v Commission, judgment of 13 February 2003 .38, 140n, 159n C-457/00 Belgium v Commission (Verlipack) [2003] ECR I-6931 .52, 120-1 C-5/01 Belgium v Commission [2002] ECR I-3452 .175n, 176n C-13/01 Safalero Srl v Prefetto di Genova [2003] ECR I-8679 .459n C-18/01 Arkkitehtuuritoimisto Riitta Korhonen Oy, Arkkitehtitoimisto Pentti Toivanen Oy,
C-327/00 Santex SpA v Unità Socio Sanitaria and others [2003] ECR I-1877 .403n T-366/00 Scott v Commission [2007] ECR II-797 .393n T-369/00 Département du Loiret (France) v Commission [2007] ECR II-851 .393n C-373/00 Adolf Truley GmbH v Bestattung Wien GmbH [2003] ECR-193 .164n C-409/00 Spain v Commission, judgment of 13 February 2003 .38, 140n, 159n C-457/00 Belgium v Commission (Verlipack) [2003] ECR I-6931 .52, 120-1 C-5/01 Belgium v Commission [2002] ECR I-3452 .175n, 176n C-13/01 Safalero Srl v Prefetto di Genova [2003] ECR I-8679 .459n C-18/01 Arkkitehtuuritoimisto Riitta Korhonen Oy, Arkkitehtitoimisto Pentti Toivanen Oy,

C-91/01 Italy v Commission (aid for small enterprises) [2004] ECR I-4355
C-96/01 P Galileo v Council [2002] ECR I-4025
T-109/01 Fleuren Compost BV v Commission, 14 January 2004
Joined Cases T-116/01 and T-118/01 P & O European Ferries (Vizcaya) et Diputación Foral de Vizcaya v Commission [ECR] 2003, II-2957
C-126/01 Ministre de l'Économie, des Finances et de l'Industrie v GEMO SA [2003] ECR I-13769 85, 117
C-159/01 Netherlands v Commission (exemptions from levies for indoor farming) [2004] ECR I-4461
T-176/01 Ferriere Nord v Commission [2004] ECR II-3931
Joined Cases T-195/01 R and T-207/01 R, Government of Gibraltar v Commission [2001] ECR II-3915
T-198/01 R Technische Glaswerke Ilmenau v Commission [2002] ECR II-2153
Joined Cases T-227–229, 265, 266, and 270/01 Diputación Foral de Álava e.a. v Commission [2009] ECR II-3029
T-252/01 Commission v Belgium [2003] ECR I-11859
Joined Cases C-261/01 and C-262/01 van Calster and Cleeren ECLI:EU:C:2003:571 230, 451nn, 452
Joined Cases C-264/01, C-306/01, C-354/01, and C-355/01 AOK Bundesverband et al. v Ichthyol- Gesellschaft Cordes, Hermani & Co. et al. [2004] ECR I-2524
C-276/01 Steffensen [2003] ECR I-3735
T-301/01 Alitalia v Commission [2008] ECR II-1753
C-486/01 P National Front v Parliament [2004] ECR I-6289
T-5/02 Tetra Laval v Commission [2002] ECR II-438158-
T-34/02 Le Levant and Others v Commission [2006] ECR II-267
C-37/02 Van Parys v Belgische Interventie- en Restitutiebureau [2005] ECR I-1465
C-66/02 Italy v Commission [2005] ECR I-10901
C-93/02 P Biret International v Council [2003] ECR I-10497
C-110/02 Commission v Council (Portuguese pig producers) [2004] ECR I-6333
C-141/02 P Commission v max.mobil [2005] ECR I-1283581
C-170/02 Schlüsselverlag J.S. Moser and Others v Commission [2003] ECR I-9889
T-171/02 Regione autonoma della Sardegna v Commission [2005] ECR II-0000
C-174/02 Streekgewest [2005] ECR I-85
Joined Cases C-183/02 P Daewoo Electronics Manufacturing España SA and C-187/02 P Territorio
Histórico de Álava [2004] ECR I-10609
Joined Cases C-186/02 and C-188/02 Territorio Histórico de Álava—Diputación Foral de Álava, Ramondín, SA e Ramondín Cápsulas, SA v Commission [2003] ECR I-24151321
Joined Cases C-189/02 P, C-202/02 P, C-205/02 P to C-208/02 P, and C-213/02 P Dansk Rørindustri and Others v Commission of the European Communities [2005] ECR I-5425
T-217/02 Ter Lembeek International NV v Commission [2006] ECR II-4483
T-237/02 R Technische Glaswerke Ilmenau v Commission [2006] ECR II-5131
C-276/02 Spain v Commission [2004] ECR I-8091
T-297/02 ACEA v Commission ECR, ECLI:EU:T:2009:189
T-300/02 AMGA v Commission [2009] ECR II-1737
T-309/02 Acegas v Commission [2009] ECR II-1809
C-345/02 Pearle HV, Hans Prijs Optiek Franchise BV and Rinck Opticiëns BV v Hoofdbedrijfschap Ambachten
C-351/02 Deutsche Bahn AG v Commission [2006] ECR II-1047
T-357/02 RENV Freistaat Sachsen v Commission [2011] ECR II-5415
T-361/02 Deutsche Bahn AG v Commission ECLI:EU:T:2005:244

C-12/03 P Commission v Tetra Laval [2005] ECR I-9875	57n, 58n, 60n, 437-8
T-17/03 Schmitz-Gotha Fahrzeugwerke v Commission of the European Communities [2 ECR II-1139	
T-20/03 Kahla/Thüringen Porzellan v Commission [2008] ECR II-2305	
C-26/03 Stadt Halle, RPL Recyclingpark Lochau GmbH v Arbeitsgemeinschaft Thermis Restabfall- und Energieverwertungsanlage TREA Leuna [2005] ECR I-1	che
T-28/03 Holcim (Deutschland) v Commission [2005] ECR II-1357	
T-30/03 RENV 3F v Commission [2011] ECR II-6651	
C-78/03 P Commission v Aktionsgemeinschaft Recht und Eigentum [2005] ECR I-10741	
C-88/03 Portugal v Commission (Azores) [2006] ECR I-7115	
T-95/03 AAESCAM v Commission [2007] ECR 1-10515	
C-131/03 P Reynolds Tobacco and Others v Commission [2006] ECR I-7795	
Joined Cases C-138/03, C-324/03, and C-431/03 Italy v Commission [2005] ECR I-1004	
C-141/03 Sniace v Commission [2005] ECR II-1197	
C-172/03 Wolfgang Heiser v Finanzamt Innsbruck [2005] ECR I-1627	
Joined Cases C-182/03 and C-217/03 Belgium and Forum 187 v Commission [2006] ECR I-5479	
T-189/03 ASM Brescia v Commission [2009] ECR II-1831	
C-205/03 Federación Española de Empresas de Tecnología Sanitaria (FENIN) v Commi [2006] I-6319	ission
T-212/03 MyTravel Group v Commission [2008] ECR II-1967	
T-271/03 Deutsche Telekom AG v Commission [2008] ECR II-477	
C-276/03 Scott v Commission [2005] ECR 1-8437	
T-285/03 Agraz and Others v Commission [2005] ECR II-1063	
T-289/03 BUPA and Others v Commission [2008] ECR II-81	
T-300/03 Moser Baer India v Council [2006] ECR II-3911	
T-301/03 Italy v Commission [2005] ECR I-10217	
T-328/03 O2 (Germany) GmbH & Co OHG v Commission [2006] ECR II-1231	
Joined Cases C-346/03 and C-529/03 Atzeni, Scalas, Lilliu and Others v Regione autono	
Sardegna [2006] ECR I-1875	238, 389n, 391n, 436
T-348/03 Koninklijke Friesland Foods v Commission [2007] ECR II-101	390 <i>n</i>
T-349/03 Corsica Ferries France v Commission [2005] ECR II-2197	299, 439 <i>n</i>
T-351/03 Schneider Electric v Commission [2007] ECR II-2237	58 <i>n</i>
T-369/03 Arizona Chemical e.a. v Commission [2005] ECR II-5839	41n
C-379/03 P Pérez Escolar v Commission ECLI:EU:C:2004:580	430-1
T-383/03 Hynix Semiconductor v Council [2005] ECR II-621	555 <i>n</i>
T-388/03 Deutsche Post and DHL International v Commission [2009] ECR II-00199	360, 424n, 446nn
C-399/03 Commission v Council (Belgian coordination centres) [2006] ECR I-5629	223, 450–1 <i>n</i>
C-415/03 Commission v Greece [2005] ECR I-3875	34n, 385n, 386n, 395
T-442/03 SIC v Commission [2008] ECR II-1161	71n, 430, 442–3n
Joined Cases C-485/03 to C-490/03 Commission v Spain [2006] ECR I-11887	396n
T-16/04 Arcelor v Parliament and Council [2010] ECR II-211	432n
T-25/04 González y Díez, SA v Commission [2007] ECR II-3121	54
C-39/04 Laboratoires Fournier SA contre Direction des verifications nationales et interi [2005] ECR I-02057	
C-71/04 Administración del Estado v Xunta de Galicia [2005] ECR I-7419	159
T-81/04 Bouygues and Bouygues Télécom v Commission ECLI:EU:T:2005:41	430
C-148/04 Unicredito Italiano SpA v Agenzia delle Entrate [2005] ECR I-11137	151, 153, 155, 436
Joined Cases C-154/04 and C-155/04 Alliance for Natural Health [2005] ECR I-6451	43n
T-167/04 Askletios Kliniken v Commission [2007] ECR II-2379.	430-1, 447n

T-196/04 Ryanair v Commission [2008] ECR II-3643
Joined Cases T-211/04 and T-215/04 Government of Gibraltar and UK v Commission [2008] ECR II-3745
C-222/04 Ministero dell'Economia e delle Finanze v Cassa di Risparmio di Firenze SpA, Fondazione
Cassa di Risparmio di San Miniato and Cassa di Risparmio di San Miniato Sp [2006] ECR I-289
T-233/04 Netherlands v Commission [2008] ECR II-591
C-237/04 Enirisorse SpA v Sotacarbo SpA [2006] ECR I-02843
Joined Cases T-265/04, T-292/04, and T-504/04 Tirrenia di Navigazione and Others v Commission ECLI:EU:T:2009:48
T-291/04 Enviro Tech Europe and Enviro Tech International v Commission [2011] ECR II-8281 415n
Joined Cases C-295/04 to C-298/04 Manfredi and Others v Lloyd Adriatico Assicurazioni SpA [2006] ECR I-6619
Joined Cases T-304/04 and 316/04 Italy and Wam v Commission (Wam 1) [2006] ECR II-6155n, 156n
Joined Cases T-309/04, T-317/04, T-329/04, and T-336/04 TV 2/Denmark and Others v Commission 2008 ECR II-2935
C-310/04 Spain v Council [2006] ECR I-728559, 60, 60n
Joined Cases C-338/04, C-359/04 and C-360/04 Placanica and Others [2007] ECR I-1891
T-360/04 FG Marine v Commission ECLI:EU:T:2007:235
C-368/04 Transalpine Ölleitung in Österreich v Finanzlandesdirektion für Tirol and Others
[2006] ECR I-9957
Joined Cases T-371 and T-394/04 British Airways v Commission [1998] ECR II-240556
C-372/04 R, on the application of Yvonne Watts v Bedford Primary Care Trust and Secretary of State for Health [2006] ECR I-4376
C-374/04 Germany v Commission [2007] ECR II-4431
T-375/04 Scheucher-Fleisch and Others v Commission [2009] ECR II-4155
Joined Cases C-393/04 and C-41/05 Air Liquide Industries Belgium SA v Ville de Seraing and Province de Liège [2006] ECR I-5293
T-395/04 Air One v Commission [2006] ECR II-1343
C-404/04 P Technische Glaswerke Ilmenau v Commission [2007] ECR I-1
C-408/04 P Commission v Salzgitter [2008] ECR I-2767
C-417/04 P Regione Siciliana v Commission [2006] ECR I-3881
Joined Cases T-425/04, T-444/04, T-450/04, and T-456/04 France and others v Commission (state telecom subsidies) [2010] ECR II-2099
Joined Cases T-427/04 and T-17/05 France and France Télécom v Commission [2009] ECR II-4315 393n
T-475/04 Bouygues and Bouygues Télécom v Commission [2007] ECR II-2097,
C-519/04 P David Meca-Medina and Igor Majcen v Commission [2006] ECR I-7006
C-525/04 P Spain v Lenzing [2007] ECR I-9947
C-526/04 Laboratoires Boiron SA v Union de recouvrement des cotisations de sécurité sociale et
d'allocations familiales (Urssaf) de Lyon [2006] ECR I-469
C-51/05 P Commission v Cantina sociale di Dolianova and Others ECR, ECLI:EU:C:2008:409432n
T-68/05 Aker Warnow Werft and Kvaerner v Commission [2009] ECR II-355
Joined Cases C-75/05 P and C-80/05 P Germany v Kronofrance SA [2008] ECR I-6619 51n, 63, 226, 421n,
422n, 436n, 438
C-119/05 Ministero dell'Industria, del Commercio e dell'Artigianato v Lucchini SpA [2007] ECR I-6199
T-135/05 Crampoli v Commission [2006] ECR II-A-2-1527
T-136/05 Salvat père & fils and Others v Commission [2007] ECR II-4063
T-211/05 Italy v Commission [2009] ECR II-2777
C-232/05 Commission v France [2006] ECR I-10071
C-243/05 P Agraz and Others v Commission [2005] ECR II-1063
T-254/05 Fachvereinigung Mineralfaserindustrie v Commission ECLI:EU:T:2007:296
C-260/05 P Sniace v Commission [2007] ECR I-10005

C-282/05 P Holcim (Deutschland) v Commission, ECR, ECLI:EU:C:2007:226
C-284/05 Commission v Finland [2009] ECR I-11705
C-295/05 Asociación Nacional de Empresas Forestales (Asemfo) v Transformación Agraria SA (Tragsa) and Administración del Estado [2007] ECR I-2999
T-299/05 Shanghai Excell M&E Enterprise and Shanghai Adeptech Precision v Council [2009] ECR II-565
C-318/05 P Commission v Germany [2007] ECR I-6957
C-326/05 P Industrias Quimicas del Vallés v Commission [2007] ECR I-6557
C-337/05 Commission v Italy (Agusta Helicopters) [2008] ECR I-7313
T-354/05 Télévision française 1 SA (TF1) v Commission [2009] ECR II-47194, 414n, 434n, 440, 442
C-362/05 P Wunenburger v Commission [2007] ECR I-4333
Joined Cases T-362/05 and T-363/05 <i>Nuova Agricast and Cofra v Commission</i> ECLI:EU:T: 2008:541
C-395/05 TAGST Draht- und Biegetechnik GmbH v Hauptzollamt Aachen [2008] ECR I-1057555
Joined Cases C-402/05 P and C-415/05 P Kadi and Al Barakaat v Council and Commission [2008] ECR I-6351
T-403/05 MyTravel Group plc v Commission [2008] ECR II-2027
T-424/05 Italy v Commission ECLI:EU:T:2009:49
C-432/05 Unibet (London) Ltd and Unibet (International) Ltd v Justitiekanslern [2007] ECR I-2271 40
T-445/05 Associazione italiana del risparmio gestito and Fineco Asset Management v Commission [2009] ECR II-289
T-455/05 Componenta v Commission [2008] ECR II-336
T-6/06 wheyco v Commission ECLI:EU:T:2007:202
C-8/06 Intertanko and Others v Secretary of State for Transport [2008] ECR I-405737
T-8/06 AB v Commission, EU:T:2009:38630
C-15/06 P Regione Siciliana v Commission [2007] ECR I-2591
T-21/06 Germany v Commission [2009] ECR II-197232
C-45/06 Reliance Industries v Council [2008] ECR II-2399554, 555
T-58/06 HALTE v Commission ECLI:EU:T:2009:125
Joined Cases T-80/06 and T-182/09 Budapesti Erőmű v Commission [2012]
C-125/06 P Commission v Infront WM [2008] ECR I-1451
T-162/06 Kronoply v Commission [2009] ECR II-1
C-176/06 P Stadtwerke Schwäbisch Hall and Others v Commission ECLI: EU:C:2007:730 422–3, 445
T-179/06 Commission v Onderzoek en Advies [2009] ECR II-64
C-199/06 Centre d'exportation du livre français (CELF) and Ministre de la Culture et de la Communication v Société internationale de diffusion et d'édition (SIDE) [2008]
ECR I-469
C-213/06 P EAR v Karatzoglou [2007] ECR I-6733
T-231/06 Kingdom of Netherlands and NOS v Commission [2010] ECR II 5993371-
C-268/06 Impact v Minister for Agriculture and Food and Others [2009] ECR I-289
Joined Cases T-273/06 and T-279/06 ISD Polska v Commission [2009] ECR II-2181
T-291/06 Operator ARP v Commission ECR, ECLI:EU:T:2009:235
T-332/06 Alcoa Trasformazioni v Commission ECLI:EU:T:2009:79
Joined Cases C-341/06 and C-342/06 Chronopost and La Poste v UFEX and Others [2008]
ECR I-4777, ECLI:EU:C:2008:375
T-369/06 Holland Malt v Commission [2009]
C-390/06 Nuova Agricast Srl v Ministero delle Attività Produttive [2008] ECR I-2577 225, 228, 435, 444
C-411/06 Sogelma v AER [2008] ECR I-2771
C-419/06 Commission v Greece [2008] ECR I-27
Joined Cases C-428/06 to C-434/06 UGT-Rioja v Juntas Generales del Territorio Histórico de
Vizcaya and Others [2008] I-06747

C-441/06 Commission v France [2007] ECR 1-8887
C-445/06 Danske Slagterier v Bundesrepublik Deutschland [2009] ECR I-2119
C-480/06 Commission v Germany [2009] ECR I-4747
C-487/06 P British Aggregates v Commission [2008] I-10515 136–7nn, 230n, 421n, 422, 435, 442n
C-494/06 P Commission v Italy and Wam SpA (Wam 2) [2009] ECR I-3639 151, 152, 152n, 155n, 157, 441–2n, 444nn
C-500/06 Corporación Dermoestética v To Me Group Advertising Media [2008] ECR I-5785148n
C-521/06 P Athinaïki Techniki v Commission [2008] ECR I-5829
C-535/06 Moser Baer India v Council [2009] ECR I-7501
C-25/07 Iride and Others v Commission [2009] ECR II-245
C-47/07 P Masdar (UK) v Commission [2008] ECR I-9761
$E-50/07\ Private\ Barnehagers\ Land for bund\ v\ EFTA\ Surveillance\ Authority\ EFTA\ Ct\ Rep\ [2008]\ p\ 62.\ldots.\ 8000000000000000000000000000000000000$
Joined Cases T-81/07 to T-83/07 Maas and Others v Commission [2009] ECR II-2411 386n, 444n
C-113/07 P SELEX Sistemi Integrati v Commission [2009] ECR I-2207
C-139/07 P Commission v Technische Glaswerke Ilmenau [2010] ECR I-5885
C-169/07 Hartlauer Handelsgesellschaft mbH v Wiener Landesregierung and Others [2009]
I-01721
T-183/07 Poland v Commission [2009] ECR II-nyr
C-204/07 P C.A.S. SpA v Commission [2008] ECR I-6135
C-214/07 Commission v France [2008] ECR I-8357
T-263/07 P Estonia v Commission [2009] ECR II-3463
T-288/07 Alcan France v Commission [2007] OJ C 235
C-290/07 P Commission v Scott [2010] ECR I-776340n, 41-2n, 44n, 131n, 345n, 392, 435, 437n, 442-3n
C-300/07 Hans & Christophorus Oymanns GbR, Orthopädie Schuhtechnik v AOK Rheinland/ Hamburg [2009] ECR I-4779164n
C-308/07 P Gorostiaga Atxalandabaso v Parliament [2009] ECR I-1059,
C-319/07 P 3F v Commission [2009] ECR I-5963
C-324/07 Coditel Brabant SA v Commune d'Uccle, Région de Bruxelles-Capitale [2009] ECR I-8457166n
C-333/07 Regie Networks v Rhone Alpes Bourgogne [2008] ECR I-10807257, 336, 441–2nn, 443n, 444n
C-334/07 P Commission v Freistaat Sachsen [2008] ECR I-9465
C-349/07 Sopropé - Organizações de Calçado Lda v Fazenda Pública [2008] ECR I-10369
C-350/07 Kattner Stahlbau GmbH v Maschinenbau- und Metall- Berufsgenossenschaft [2009] ECR I-1513
C-369/07 Commission v Greece [2009] ECR I-5703
C-370/07 Commission v Council [2009] ECR I-8917
C-384/07 Wienstrom GmbH v Bundesminister für Wirtschaft und Arbeit [2008] ECR I-10393350
T-421/07 Deutsche Post v Commission [2011] ECR II-08105
T-422/07 Djebel—SGPS v Commission ECLI:EU:T:2012:11
T-423/07 Ryanair v Commission [2011] ECR II-2397
C-431/07 Bouygues and Bouygues Télécom v Commission [2009] ECR I-2665
C-441/07 P Commission v Alrosa [2010] ECR I-5949
T-442/07 Ryanair v Commission ECLI:EU:T:2009:325
Joined Cases C-445 and C455/07 P Commission v Ente per le Ville Vesuviane and Ente per le
Ville Vesuviane v Commission [2009] ECR I-7993
C-460/07 Sandra Puffer v Unabhängiger Finanzsenat, Außenstelle Linz [2009] ECR I-325168n,71
T-475/07 Dow AgroSciences and Others v Commission [2011] ECR II-5937
C-504/07 Associação Nacional de Transportadores Rodoviários de Pesados de Passageiros (Antrop)
and Others v Conselho de Ministros and Others [2009] ECR I-3867
[2010] ECR I-8533
C-519/07 P Commission v Koninklijke FrieslandCampina NV [2009] ECR I-8495
C-536/07 Commission v Germany (Köln Messe) FCR L-10355

C-2/08 Amministrazione dell'Economia e delle Finanze v Fallimento Olimpiclub [2009] ECR 1-7501 403n
Joined Cases C-78/08 and C-80/08 Paint Graphos (Italian Cooperative Societies) [2011] ECR I-07611
C-89/08 P Commission v Ireland and Others [2009] ECR I-11245
C-118/08 Transportes Urbanos y Servicios Generales [2010] ECR I-635
C-168/08 Dimosia Epicheirisi Ilektrismou AE (DEI) v Commission, Judgment of 20 September 201294-5
C-169/08 Presidente del Consiglio dei Ministri v Regione Sardegna [2009] ECR I-10821,145
T-189/08 Forum 187 v Commission [2010] ECR II-1039
T-266/08 Deutsche Post v Commission [2008] ECR II-1233
Joined Cases T-268/08 and T-281/08 Land Burgenland and Austria v Commission
C-279/08 P Commission v the Netherlands (NoX, emission trading) [2011] ECR I-7671 69-70, 136, 136-7nn, 140n, 157n, 230n
T-282/08 Grazer Wechselseitige Versicherung v Commission ECLI:EU:T:2012:91
T-304/08 Smurfit Kappa v Commission [2012] ECR II-0000
Joined Cases C-317/08 to C-320/08 Alassini and Other s v Telecom Italia SpA and Others [2010] ECR I-2213
T-335/08 BNP Paribas and BNL v Commission [2010] ECR II-03323
C-355/08 P WWF-UK v Council ECLI:EU:C:2009:286
Commission [2011] ECR II-6255
$C-399/08\ Commission\ v\ Deutsche\ Post\ [2010]\ ECR\ I-7831, ECLI: EU: C: 2010: 481\ \dots \\ 127, 364n, 374n$
$C-406/08\ Uniplex\ (UK)\ Ltd\ v\ NHS\ Business\ Services\ Authority\ [2010]\ ECR\ I-817\402-3n$
Joined Cases T-443/08 and T-455/08 Freistaat Sachsen and Land Sachsen-Anhalt (T-443/08) and Mitteldeutsche Flughafen AG and Flughafen Leipzig-Halle GmbH (T-455/08) v Commission [2011] ECR II-131981-2, 338, 413-14, 416-17, 416n
C-451/08 Helmut Müller v Bundesanstalt für Immobilienaufgaben [2010] ECR I-2673168, 168n
T-468/08 AES-Tisza v Commission [2014] ECLI:EU:T:2014:235
T-481/08 Alisei v Commission [2010] ECR II-117
Joined Cases T-494/08 to T-500/08 and T-509/08 Ryanair v Commission [2010] ECR II-05723376
C-506/08 P Sweden v My Travel and Commission [2011] ECR I-06237
C-507/08 Commission v Slovakia [2010] ECR I-13489
T-525/08 Poste Italiane v Commission ECLI:EU:T:2013:481
T-532/08 Norilsk Nickel and Others v Commission [2010] ECR II-3959
T-539/08 Etimine and Others v Commission [2010] ECR II-4017
C-542/08 Barth v Bundesministerium für Wissenschaft und Forschung [2010] ECR I-3189403n
Joined Cases T-568/08 and T-573/08 Métropole télévision (M6) and Télévision française 1 SA (TF1) v Commission [2010] ECR II-3397
T-570/08 RENV Deutsche Post v Commission [2013] ECR II-0000
T-571/08 Germany v Commission, 14 July 2010
T-579/08 Eridana Sadam v Commission [2011] ECR II-0000
C-67/09 P Nuova Agricast Srl and Cofra Srl v Commission [2010] ECR I-9811
Joined Cases C-71/09 P, C-73/09 P, and C-76/09 P Comitato 'Venezia vuole vivere' e.a. v Commission
[2011] I-4727
C-83/09 Commission v Kronoply and Kronotex [2011] ECR I-4441 45n, 345n, 422–4, 423n, 445–6
T-88/09 Idromacchine and Others v Commission [2011] ECR II-7833
T-89/09 Pollmeier Massivholz v Commission ECR (Extracts), EU:T:2015:153
Joined Cases C-106/09 P and C-107/09 P Commission and Spain v Government of Gibraltar and UK 2011 [ECR] I-11113
480, 480n, 505 Joined Cases T-115/09 and T-116/09 Electrolux AB and Whirlpool Europe BV v Commission [2012] ECR II-0000
T-123/09 P Ryanair v Commission ECR, EU:T:2012:164

T-139/09 France v Commission ECR, ECLI:EU:T:2012:496
C-148/09 P Belgium v Deutsche Post and DHL International [2011] ECR I-08573358, 359–60, 423, 424n, 445nn, 446
C-194/09 P Alcoa v Commission ECLI:EU:C:2011:497
C-210/09 Scott and Others v Ville d'Orléans [2010] ECR I-4613
C-221/09 AJD Tuna Ltd v Direttur tal-Agrikoltura u s-Sajd and Avukat Generali [2011] ECLI:EU:C:2011:153
C-240/09 Lesoochranárske zoskupenie VLK v Ministerstvo životného prostredia Slovenskej republiky [2011] ECR I-1255
T-243/09 Fedecom v Commission ECLI:EU:T:2012:497
C-279/09 Deutsche Energiehandels- und Beratungsgesellschaft mbH (DEB) v Bundesrepublik Deutschland [2010] ECR I-13849
C-320/09 P A2A v Commission ECLI:EU:C:2011:858
C-322/09 P NDSHT v Commission [2010] ECR I-11911
C-329/09 P Iride v Commission ECLI:EU:C:2011:859
C-330/09 RapidEye v Commission ECLI:EU:T:2011:48
C-331/09 Commission v Poland [2011] ECR I-2933
T-347/09 Germany v Commission ECLI:EU:T:2013:418
C-360/09 Pfleiderer AG v Bundeskartellamt [2011] ECR I-5161
C-362/09 Athinaïki Techniki v Commission [2010] ECR I-13275
C-379/09 Italy v Commission [2012] ECLI:EU:T:2012:422
T-407/09 Neubrandenburger Wohnungsgesellschaft v Commission ECLI:EU:T:2012:1
C-437/09 AG2R Prévoyance v Beaudout Père et Fils SARL [2011] ECR I-1003
Joined Cases C-444/09 and C-456/09 Gavieiro (C-444/09) and Torres (C-456/09) v Consellería de Educación e Ordenación Universitaria de la Xunta de Galicia [2010] ECR I-14031
T-457/09 Westfälisch-Lippischer Sparkassen- und Giroverband v Commission ECLI:EU:T: 2014:683
C-464/09 P Holland Malt v Commission [2010] I-12443
Joined Cases C-465/09 P and C-470/09 P Territorio Histórico de Vizcaya v Commission [2011] ECR I-0000
C-480/09 P AceaElectrabel Produzione v Commission [2010] ECR I-13355392
C-494/09 Bolton Alimentari SpA v Agenzia delle Dogane - Ufficio delle Dogane di Alessandria [2011] ECR I-647
C-496/09 P Commission v Italy [2011] ECR I-11483
T-520/09 TF1 and Others v Commission, ECLI:EU:T:2012:352
C-529/09 Commission v Spain (CJEU, 24 January 2013)
C-544/09 P Germany v Commission [2011] ECR I-128232
C-550/09 Criminal proceedings against E and F [2010] ECR I-6213
T-18/10 Inuit Tapiriit Kanatami and Others v Parliament and Council [2011] ECR II-5599
Joined Cases T-29/10 and T-33/10 <i>Netherlands and ING Groep v Commission</i> , Judgment of 2 March 2012, ECLI:EU:T:2012:98
F-42/10 Skareby v Commission [2012] ECR nyr of 16 May 2012
C-47/10 P Austria v Scheucher-Fleisch and Others [2011] ECR I-10707423, 423–4nn, 445–6, 445n, 447
C-69/10 Samba Diouf v Ministre du Travail, de l'Emploi et de l'Immigration [2011] ECR I-7151
T-79/10 Colt Télécommunications France v Commission, Judgment of 16 September 2013
C-81/10 P France Télécom SA v Commission [2011] ECR I-12941
C-111/10 Commission v Council (Lithuania)
C-117/10 Commission v Council (Poland)
C-118/10 Commission v Council (Latvia)

C-121/10 Commission v Council (Hungary)	223n
C-124/10 P Commission v Électricité de France (EDF) and Others, Judgment of 5 June not yet published	
T-137/10 Coordination bruxelloise d'institutions sociales et de santé (CBI) v Commissi not yet published	ion [2012]
T-154/10 France v Commission ECLI:EU:T:2012:452	413n, 414n, 441n
T-156/10 Confederación de Cooperativas Agrarias de España and CEPES v Commissio ECLI:EU:T:2014:41	on 415n
T-182/10 Aiscat v Commission, judgment of 15 January 2013	
T-219/10 Autogrill España, SA v Commission [2014]	
T-221/10 Iberdrola v Commission ECLI:EU:T:2012:112	
T-234/10 Ebro Foods v Commission ECLI:EU:T:2012:141	
T-236/10 Asociación Española de Banca v Commission ECLI:EU:T:2012:176	425n, 427n
C-243/10 Commission v Italy (CJEU, 29 March 2012)	384n
T-257/10 Italy v Commission ECLI:EU:T:2012:504	
T-262/10 Microban and Others v Commission [2011] ECR II-7697	
C-275/10 Residex Capital [2011] ECR I-13043	
T-303/10 Wam Industriale v Commission ECLI:EU:T:2012:505	
T-344/10 P UPS Europe and United Parcel Service Deutschland v Commission ECLI:E 2012:216	U:T:
T-362/10 Vtesse Networks v Commission ECLI:EU:T:2014:928	
C-386/10 P Chalkor v Commission ECLI:EU:C:2011:815.	
C-390/10 P Füller-Tomlinson v EP (GC, 6 December 2012)	390 <i>n</i>
Joined Cases C-399/10 P and 401/10 P Bouygues SA and Bouygues Telecom SA v Command Others, judgment of 19 March 2013	mission
C-403/10 P Mediaset SpA v Commission [2011] ECR I-117	
C-417/10 Ministero dell'Economia e delle finanze v 3M Italia SpA [2012] ECLI:EU:C:2	
C-451/10 P TF1 v Commission ECLI:EU:C:2011:384	
C-452/10 BNP Paribas and BNL v Commission [2010] ECR II-03323	
C-459/10 P Freistaat Sachsen and Others v Commission (2011), not yet reported	
Joined Cases C-463/10 P and C-475/10 P Deutsche Post and Germany v Commission ECR 1-0000	[2011]
T-499/10 MOL Magyar Olaj- és Gázipari Nyrt v Commission [2013]	
T-551/10 Fri-El Acerra v Commission ECLI:EU:T:2013:430	\dots 260, 438 n , 441 n
T-555/10 JBF RAK v Council, judgment of 24 May 2012	
T-556/10 Novatex v Council, judgment of 11 October 2012	
C-610/10 Commission v Spain (CJEU, 11 December 2012)	
T-15/11 Sina Bank v Council ECLI:EU:T:2012:661	
T-57/11 Castelnou Energia v Commission [2014] ECR II-0000	
C-73/11 Frucona Košice v Commission (24 January 2013) ECLI:EU:C:2013:32 12	
C-138/11 Compass-Datenbank GmbH v Republik Österreich, Judgment of 12 July 201	2 77, 79, 81
T-151/11 Telefónica de España and Telefónica Móviles España v Commission, EU:T:20	014:631 304, 437 <i>n</i>
C-167/11 Cantiere navale De Poli v Commission [2012] ECR I-0000	227
T-174/11 Modelo Continente Hipermercados v Commission ECLI:EU:T:2012:143	427n
C-226/11 Expedia Inc v Autorité de la concurrence and Others [2012] not yet publishe	d 25
Joined Cases C-239/11 P, C-489/11 P and C-498/11 P Siemens and Others v Commissi ECLI:EU:C:2013:866	
Joined Cases C-247/11 P and C-253/11 P Areva v Commission ECLI:EU:C:2014:257	434n
T-251/11 Austria v Commission [2014] ECR II-0000	228-9 <i>n</i>
C-262/11 Kramikovtzi (CJEU, 29 November 2012)	351, 352
T-275/11 Télévision française 1 (TF1) v Commission [2013] ECR II-0000	229

C-277/11 M.M. v Minister for Justice, Equality and Law Reform, Ireland, and Attorney General ECLI:EU:C:2012:744
C-288/11 P Mitteldeutsche Flughafen AG and Flughafen Leizig-Halle GmbH v Commission,
Judgment of 19 December 2012, ECLI:EU:C:2012:821
T-319/11 ABN Amro Group NV v Commission, ECLI:EU:T:2014:186
T-381/11 Eurofer v Commission ECLI:EU:T:2012:273
T-385/11 BP Products North America v Council, judgment of 16 January 2014
T-399/11 Banco Santander SA and Santusa Holding SL v Commission [2014]
T-400/11 Altadis v Commission ECLI:EU:T:2013:490
T-444/11 Gold East Paper and Gold Huasheng Paper v Council (pending)555n
T-477/11 P Sepracor Pharmaceuticals v Commission ECLI:EU:C:2012:292
T-487/11 Banco Privado Portugues and Others v Commission (GC, 12 December 2014) 383n, 389n, 436n, 438nn
T-488/11 Scheepsbouwkundig Advies-en Rekencentrum (SARC) BV v Commission ECLI:EU:T:
2014:497
T-489/11 Rousse Industry v Commission ECLI:EU:T:2013:144
C-501/11 P Schindler Holding and Others v Commission ECR, EU:C:2013:522
C-509/11 ÖBB-Personenverkehr ECLI:EU:C:2013:61
C-510/11 P Kone and Others v Commission ECLI:EU:C:2013:696
T-512/11 Ryanair v Commission ECLI:EU:T:2014:989
C-583/11 P Inuit Tapiriit Kanatami and Others v Parliament and Council ECLI:EU:C: 2013:625
T-601/11 Dansk Automat Brancheforening v Commission ECLI:EU:T:2014:839
C-613/11 Commission v Italy (CJEU, 21 March 2013)
C-615/11 P Commission v Ryanair ECLI:EU:C:2013:310
Joined Cases C-630/11 P to C-633/11 P HGA and Others v Commission (Sardinian hotel industry), ECLI:EU:C:2013:387
C-677/11 Doux Élevages and Coopérative agricole UKL-AREE v Ministère de l'Agriculture,
judgment of 30 May 2013
Joined Cases T-15/12 and T-16/12 Provincie Groningen and Others v Commission ECLI:EU:T: 2013:74
C-34/12 P Idromacchine and Others v Commission ECLI:EU:C:2013:552
T-52/12 Greece v Commission [2014] ECR II-0000
C-77/12 P Deutsche Post AG v European Commission
T-79/12 Cisco Systems and Others v Commission, 11 December 2013
Joined Cases C-105/12 to C-107/12 The Netherlands v Essent NV (C-105/12), Essent Nederland BV (C-105/12), Eneco Holding NV (C-106/12) and Delta NV (C-107/12), Judgment of 22 October 2013
C-132/12 P Stichting Woonpunt and Others v Commission ECLI:EU:C:2014:100
C-133/12 P Stichting Woonlinie and Others v Commission ECLI:EU:C:2014:105
T-150/12 Greece v Commission ECLI:EU:T:2014:191
Joined Cases C-214/12P, C-215/12P, and C-223/12 P Land Burgenland and Others v Commission,
Judgment of 24 October 2013
C-224/12 P Commission v Netherlands and ING Groep ECLI:EU:C:2014:213
C-248/12 P Northern Ireland Department of Agriculture and Rural Development v Commission ECLI:EU:C:2014:137
C-262/12 Association Vent De Colère! Fédération nationale and Others v Ministre de l'Écologie, du Développement durable, des Transports et du Logement and Ministre de l'Économie, des Finances et de l'Industrie, judgment of 19 December 2013
C-263/12 Commission v Greece (CJEU, 17 October 2013)
C-272/12 P Commission v Ireland and Others ECLI:EU:C:2013:812
C-274/12 P <i>Telefonica SA v Commission</i> judgment of 19 December 2013
428-9. 428n

C-276/12 Sabou v Financní reditelství pro hlavní mesto Prahu (CJEU, 22 October 2013)43n
C-284/12 Lufthansa v Frankfurt-Hahn (CJEU, 21 November 2013) 342, 350, 383n, 388n, 405n, 410, 455-6
C-287/12 P Ryanair v Commission ECLI:EU:C:2013:395
T-295/12 Germany v Commission [2014] ECR II-0000
T-309/12 Zweckverband Tierkörperbeseitigung v Commission [2014] ECR II-0000 224, 227, 436n, 437n
Joined Cases T-319/12 and T-321/12 Spain and Others v Commission, EU:T:2014:604307
C-362/12 Test Claimants in the Franked Investment Income Group Litigation v Commissioners of Inland Revenue ECLI:EU:C:2013:834
C-370/12 Pringle v Government of Ireland, Ireland and Attorney General CLI:EU:C:2012:756 $\dots \dots 405n$
T-413/12 Post Invest Europe v Commission ECLI; EU:T:2013:246
T-461/12 Hansestadt Lübeck v Commission ECLI:EU:T:2014:758 (appeal C-524/14 P pending) 410n, 447n
T-499/12 HSH Investment Holdings Coinvest-C Sàrl v Commission, EU:T:2015:840276, 277-8
T-500/12 Ryanair v Commission ECLI:EU:T:2015:73
C-517/12 Alro v Commission ECLI:EU:T:2014:890
C-527/12 Commission v Germany (CJEU, 11 September 2014)
Joined Cases C-533/12 P and C-536/12 P SNCM and France v Corsica Ferries France ECR, ECLI:EU:C:2014:2142
T-545/12 Mory and Others v Commission ECLI: EU:T:2013:607
C-599/12 P France v Commission (CJEU, 13 March 2014)
C-604/12 H.N. v Minister for Justice, Equality and Law Reform (CJEU, 8 May 2014)
C-27/13 Flughafen Lübeck v Air Berlin plc & Co. Luftverkehrs-KG ECLI:EU:C:2014:240350
C-57/13 Club Hotel Loutraki and Others v Commission ECLI:EU:T:2014:183,
C-69/13 Mediaset SpA v Ministerio dello Sviluppo Economico (CJEU, 13 February 2014) 383n, 384, 386–8
T-129/13 Alpiq RomIndustries and Alpiq RomEnergie v Commission ECLI:EU:T:2014:895
T-162/13 Magic Mountain v Commission ECLI:EU:T:2016:341
C-242/13 Commerz Nederland N.V. v Rotterdam Port Authority (HbR), judgment of 17 September 2014
T-251/13 Gemeente Nijmegen v Commission ECLI:EU:T:2015:142
C-303/13 P Andersen v Commission [2015] ECR I-0000
T-321/13 Adorisio and Others v Commission ECLI:EU:T:2014:175
C-456/13 P T & L Sugars and Sidul Açúcares v Commission ECR, EU:C:2015:284
C-518/13 R, on the Application of Eventech Ltd v The Parking Adjudicator, judgment of 14 January 2015
Joined Cases C-587/13 P and C-588/13 P Banco Bilbao Vizcaya Argentaria and Telefónica v Commission, EU:C:2015:18
C-667/13 Banco Privado Português and Massa Insolvente do Banco Privado Português ECLI:EU:C:2015:151405n, 424, 444n, 452n
$C-672/13\ OTP\ Bank\ Nyrt\ v\ Magyar\ \'Allam\ and\ Magyar\ \'Allamkincst\'ar\ EU: C: 2015: 185403n, 452, 454, 459$
C-690/13 Trapeza Eurobank Ergasias AE [2015] ECR I-00000
C-15/14 MOL Magyar Olaj- és Gázipari Nyrt v Commission [2013]131–2n
$C-33/14\ P\ Mory\ and\ Others\ v\ Commission\ ECLI: ECR$
$T-40/14\ Electrabel\ and\ Dunamenti\ Er\"{o}m\"{u}\ v\ Commission\ ECLI: EU: T:2014: 1004\$
$C-505/14\ Klausner\ Holz\ Niedersachsen\ v\ Land\ Nordrhein-Westfalen\ EU: C:2015:742\\\ .402-3n$
C-93/15 P Banco Privado Português and Massa Insolvente do Banco Privado Português v Commission ECLI:EU:C:2015:703
INTERNATIONAL CASES
European Court of Human Diabte (ECtUD)
European Court of Human Rights (ECtHR) Jussila v Finland, Application no 73053/01 (ECtHR, 23 November 2006)
Kudla v Poland, Application no 30210/96 [General Court] ECHR 2000-XI

M.S.S. v Belgium and Greece, Application no 30696/09[General Court] ECHR 2011
Ombudsman 371n Case 1184/2012 (ER) PMC of 24 April 2014
World Trade Organization
Australia—Subsidies provided to producers and exporters of automotive leather, Panel Report, DS 126 491
Brazil—Export Financing Programme for Aircraft, Appellate Body Report, WT/DS46/AB/R
Canada—Certain measures affecting the renewable energy generation sector (Canada—Renewable Energy), Appellate Body Report, WT/DS412/AB/R
Canada—Measures affecting the importation of milk and the exportation of dairy products, Panel Report, WT/DS103/R of 17 May 1999
Canada—Measures affecting the importation of milk and the exportation of dairy products (Article 21.5—New Zealand and US II), Appellate Body Report, WT/DS113/AB/R214–15
Canada—Measures relating to the feed-in tariff program, Appellate Body Report, WT/DS426/AB/R 473
Canada—Measures relating to the feed-in tariff program (Canada—FIT), Appellate Body Report, WT/DS426/AB/R
China—Countervailing and anti-dumping duties on grain oriented flat-rolled electrical steel from the United States (China—GOES), Panel Report, WT/DS414/R
European communities and certain Member States—Measures affecting trade in large civil aircraft, Panel Report, WT/DS316/R
European communities and certain Member States—Measures affecting trade in large civil aircraft, Appellate Body Report, WT/DS316/AB/R
European communities—Anti-dumping duties on import of cotton-type bed linen from India (Article 21.5), Appellate Body Report, WT/DS141/AB/R
European communities—Countervailing measures on dynamic random access memory chips (DRAMS) from Korea, Panel Report, WT/DS299/R
$\label{thm:communities} \textit{European communities} - \textit{Export subsidies on sugar}, \textit{Panel Report, WT/DS265/R} \dots 214$
$\textit{European communities} - \textit{Export subsidies on sugar}, \\ \textit{Appellate Body Report, WT/DS265/AB/R} \dots 214-15, 507$
$\textit{European communities} - \textit{Measures affecting trade in commercial vessels}, Panel Report, WT/DS301/R \dots 507$
Indonesia—Certain measures affecting the automotive industry, Panel Report, WT/DS54, 55, 59, 64/R
$\textit{Philippines-Taxes on distilled spirits}, Appellate Body Report, WT/DS396 and 403/AB/R. \dots 497$
United States—Anti-Dumping and Countervailing Duties (China), Appellate Body Report, WT/DS379/AB/R
United States—Countervailing duties and anti-dumping measures on certain products from China, Appellate Body Report, WT/DS/449/AB/R
United States—Countervailing duty investigation on Dynamic Random Access Memory Semiconductors (DRAMS) from Korea, Appellate Body Report, WT/DS296/AB/R
United States—Countervailing duty measures on certain products from China, Appellate Body Report, WT/DS437/AB/R
United States—Countervailing measures concerning certain products from the European communities, Appellate Body Report, WT/DS212/AB/R
United States—Countervailing Measures on Certain Hot-Rolled Carbon Steel Flat Products from India, Appellate Body Report, WT/DS436/AB/R
United States—Definitive anti-dumping and countervailing duties on certain products from China, Appellate Body Report, WT/DS379/AB/R
United States—Final countervailing duty determinations with respect to certain softwood lumber from Canada (US—Softwood Lumber IV), Appellate Body Report, WT/DS257/AB/R478, 482, 489
United States—Imposition of countervailing duties on certain hot-rolled lead and bismuth carbon steel products originating in the United Kingdom, Appellate Body Report, WT/DS138/AB/R 487

United States—Measures affecting trade in large civil aircraft (second complaint), P	anel Report,
WT/ DS353/R	477-8, 480-1, 482
United States—Measures affecting trade in large civil aircraft (second complaint), A	ppellate Body
Report, WT/ DS353/AB/R	477–8, 481 <i>n</i> , 493 <i>n</i>
${\it United States-Measures treating export restraints as subsidies}, Panel Report, WT/WT/WT/WT/WT/WT/WT/WT/WT/WT/WT/WT/WT/W$	T/DS194/R476, 478
United States—Tax treatment for 'foreign sales corporations'—Recourse to Article 2	21.5 of the DSU
by the European Communities (US-FSC), Appellate Body Report, WT/DS10	8/AB/RW480

$Table\ of\ Legislation$

n =footnote.

EU LAW	TEU (Treaty on European Union)	
Treaties and Conventions	Art 3(3)	
Accession Treaty of Austria, Finland and	Art 4(2)62, 90	
Sweden 1994	Art 4(3)	
Charter of Fundamental Rights of the EU 343-4	Art 6(1)(1)	
Art 1140	Art 6(3)	
Art 1340	Art 740	
Art 2240	Art 840	
Art 2840	Art 940	
Art 34	Art 10	
Art 3540	Art 11	
Art 36	Art 11(3)	
Art 3740	Art 1240	
Art 41 42–3, 42 <i>n</i> , 47, 341, 374 <i>n</i> , 389	Art 13(2)	
Art 41(1)	Art 19(1)	
Art 41(2)	Art 19(1)(1)	
Art 41(3)	Art 19(1)(1)(2)401-2	
Art 42	Art 19(1)(2)	
Art 47	Protocol 26	
Art 47(1)	TFEU (Treaty on the Functioning of the	
Art 47(2)	European Union)	
Art 51(1)	Art 364	
Art 52(2)	Art 4(3) 385, 456, 456 <i>n</i> , 462	
EEC Treaty see TFEU	Art 6(3)	
Euratom Treaty (Treaty of European Atomic	Art 14 88, 88 <i>n</i> , 224	
Energy Community 1957)20, 20 <i>n</i>	Art 16 169, 169 <i>n</i> , 175–6	
European Coal and Steel Community (ECSC)	Art 30230	
Treaty 1957	Art 34230	
Art 2231	Art 3788n	
Art 3231	Art 42242	
Art 4231	Art 58(1)308	
Art 95231	Art 71310	
European Convention on Human Rights and	Art 87 (former) see Art 107	
Fundamental Freedoms 1950 (ECHR)	Art 88172 <i>n</i>	
Art 6	Art 91 (ex 75)	
Art 13	Art 92 (former)see Art 107	
European Economic Area (EEA) Agreement	Art 93 (ex 77)219, 224–5, 230, 308,	
Art 61	309–10, 315–18	
Maastricht Treaty 1992	Art 93 (former) see Art 108	
Paris Treaty Establishing the ECSC 195118–19	Art 100(1)	
Art 4(b)	Art 101	
Statute of the European Court of Justice	Art 101(1)	
Art 21(1)	Art 101(3)	
Art 46	Art 102	
Art 53(1)	Art 106 (ex 90) 88, 168–9, 170, 309–10	

TFEU (Treaty on the Functioning of the	Art 147(2)
European Union) (cont.):	Art 167 (ex 151)
Art 106(1)	Art 167 (ex 131)
Art 106(2)	Arts 174-178
94–6, 102, 103, 104, 169, 172,	Art 185
172 <i>n</i> , 178, 178 <i>n</i> , 179–80, 224–5,	Art 187
229-30, 303-5, 308, 309-10,	
314, 315, 320–1, 437,	Art 216(2)
439–40, 447	Art 263
Art 106(3)89, 91, 96–7, 309–10, 317–18	Art 263 (ex 230)
Art 107 (ex 92/87) 23, 39, 48, 65, 108, 114–16,	
172n, 461, 516–17, 516n	Art 263(1)
Art 107(1) 1, 6, 10, 49, 52 <i>n</i> , 63 <i>n</i> , 65–6, 65 <i>n</i> , 68, 70, 71, 84–5, 94–5, 105, 108–9,	Art 263(2)
124-5, 129-30, 129 <i>n</i> , 130 <i>n</i> , 131 <i>n</i> , 136-7 <i>n</i> ,	Art 263(4)
137, 140 <i>n</i> , 141 <i>n</i> , 151–3, 158–9, 163, 166,	426-7, 428-9, 430, 456, 554
172, 175–6, 178 <i>n</i> , 194, 198, 203–5, 206–7,	Art 263(6)
219, 224, 234, 282, 284, 285, 303, 335-6,	Art 264(1)
338, 388, 410, 416, 433, 435, 436, 444,	Art 265
454–5, 478, 483, 487, 490, 505	Art 265(1)
Art 107(2) 56, 141, 191, 219, 224–5, 234–9,	Art 265(2)
343, 348, 350-1, 435-6, 438, 439, 447, 516n	Art 265(3)
Art 107(2)(a)	Art 265(4)
	Art 266
Art 107(2)(b)	Art 266(1)
Art 107(2)(c) 21, 141 <i>n</i> , 149, 234–5, 246	Art 26747, 388, 405, 429, 555
Art 107(3) 21, 56–7, 77 <i>n</i> , 89, 104, 141, 141 <i>n</i> , 191, 219, 224–6, 230, 232, 234–5, 240–307,	Art 267(1)(b)405
325-6, 343, 348, 350-1, 435-6, 438, 439,	Art 267(1)(b)424
447, 452, 516n	Art 268406, 431–2
Art 107(3)(a) 21, 191–2, 194 <i>n</i> , 195, 195 <i>n</i> ,	Art 277 405 <i>n</i> , 429, 438
196-7, 225, 240, 245, 248-51, 290	Art 283
Art 107(3)(b)240, 267-8, 269, 276-8	Art 288(3)
Art $107(3)(c) \dots 191-2, 194n, 195, 195n, 196-7,$	Art 288(4)
228-9, 240, 248-51, 286, 290,	Art 288(5)
299, 315, 317, 325	Art 289(2)
Art 107(3)(d)240, 300-1, 303-5, 306-7	Art 289(3)
Art 107(3)(e)	Art 290
Art 108 (ex 93)23, 30n, 39, 48, 56, 90, 100,	Art 291
114–15, 219, 303, 313, 331, 348–9,	Art 294
361, 388, 406–7, 463	Art 296
Art 108(1) 53-4, 220, 226, 329, 348	Art 296(2)
Art 108(2) 27, 45 <i>n</i> , 46, 54, 56, 61, 115–16,	Art 304(2)
223-5, 237, 317, 360, 362, 364-5, 368, 394, 406, 419-23, 424, 431, 435, 445, 457	Art 339
	Art 340
Art 108(3) 21–2, 49, 61, 68, 159, 219–20, 224, 286, 311, 320, 325, 328, 342, 348,	
357, 382, 383, 388, 409–10, 416–17, 419,	Art 340(2)
452 <i>n</i> , 453, 454–5, 458–9, 460	Art 345
Art 108(3)(3)448	Art 346
Art 108(4)	Art 349
Art 109 (ex 94) 28, 39, 48, 220, 224, 224n,	
309, 314	Directives (in chronological order)
Art 109(4)	Directive (EC) 71/305 (Public Works Directive)
Art 110230	Art 29(4)
Art 117	Art 29(a)

Directive (EC) 77/62 (Public Supplies	Art 2(1)(b)	$164n, 165n$
Directive)	Art 2(3)	164 <i>r</i>
Art 26180 <i>n</i>	Art 21	163 <i>r</i>
Commission Directive 80/723/EEC of 25 June	Art 22(a)	163 <i>r</i>
1980 on the transparency of financial	Art 23(2)(a)	
relations between Member States and	Art 23(2)(b)	
public undertakings (OJ L 195/35 of	Explanatory Memorandum (C	
29 July 1980)	347-SYN 36 1)	
Art 5	Directive 2004/18/EC of the Euro	
Council Directive 81/363/EEC of 28 April 1981 on aid to shipbuilding (OJ L137, 23.05.1981)106	Parliament and of the Counc 2004 coordinating the proce award of contracts, OJ L134/	cil of 31 March dures for the
Directive 89/229122	Sector Directive)	
Council Directive 89/391 on the introduction	Art 1(2)(a)	
of measures to encourage improvements	Art 1(2)(b)	
in the safety and health of workers	Art 1(2)(c)	
at work		
Directive 92/57 on the implementation of	Art 1(2)(d)	
minimum safety and health requirements at temporary or mobile construction	Art 1(9)	
sites	Art 3	
Commission Directive 93/84/EEC of 30	Art 14	
September 1993 amending Directive	Art 15	
80/723/EEC, (OJ) 1993 No L 254/16 115	Art 18	
Directive 96/71/EC of the European Parliament	Directive 2004/38/EC of the Euro	*
and of the Council concerning the	Parliament and of the Counc	
posting of workers in the framework	2004 on the right of citizens their family members to mo	
of the provision of services, OJ L 18/1	freely within the territory of	
of 21/01/1997	States, OJ L 158	
Directive 96/92/EC of the European Parliament	Council Directive 2005/85/EC of	
and of the Council of 19 December 1996 concerning common rules for the internal	2005 on the minimum stand	ards on
market in electricity, OJ 1997 L 27	procedures in Member State	
Directive 98/50, OJ L 132 [1998]184 <i>n</i>	granting and withdrawing re	
Directive 2000/43/EC of 29/06/2000 implement-	status, OJ L 326/13	
ing the principle of equal treatment between	Commission Directive 2006/111/	
persons irrespective of racial or ethnic	16 November 2006 on the tra of financial relations betwee	
origin, OJ 2000 L 180/22183 <i>n</i>	States and public undertakir	
Directive 2000/78/EC of 27/11/2000 establishing	OJ 2006, L318/17	
a general framework for equal treatment in	Directive 2007/66, OJ 2007, L335/	
employment and occupation, OJ 2000	Directive)	
L 303/16	Directive 2009/81/EC, OJ L 2009	(Defence and
Directive 2001/23 on the safeguarding of	Security Procurement Direc	tive) 162
employees' rights in the event of transfers of undertakings, businesses, or parts of	Directive 2011/24/EU of the Euro	pean Parliament
undertakings or businesses, OJ L 82/16 of 22/	and of the Council of 9 Marc	
03/2001, codifying Directive	application of patients' right	
77/187/EEC	border healthcare, OJ 2011, I	.88/45101
Directive (EC) 2002/22 (Universal Services	Regulations (in chronological or	der)
Directive)	Council Regulation (EC) No 1191	
Directive 2004/17/EC of the European	protection against dumping	
Parliament and of the Council of 31 March	ing of bounties or subsidies b	
2004 coordinating the procurement procedures in the public services, OJ L134/	which are not members of the	
1 (Utilities Directive)	Economic Community, OJ 19	
Art 1(3)	Council Regulation (EC) No 1191	
Art 2(1)(a) 163n 179n	by Regulation (EC) No 1370/2	2007 100-1

Regulation (EEC) No 1192/69 on common	Council Regulation No 3577/92 applying the
rules for the normalization of the accounts	principle of freedom to provide sevices to
of railway undertakings 221, 308, 313–14	maritime transport within Member States
Recital 1	(maritime cabotage), OJ L 364 308, 319, 320
Art 2313	Art 6320
Art 4(1)	Council Regulation (EC) No 3286/94 of
Art 4(2)	22 December 1994 laying down Community
Art 4(3)	procedures in the field of the common commercial policy, OJ 1994 L349/71
Art 4(4)	(Trade Barriers Regulation)508n
Art 13(2)	Council Regulation (EC) No 3294/94 amended
Regulation (EEC) No 1107/70 on the granting	by Council Regulation 1252/95, OJ 1994
of aids for transport by rail, road and	L349/22508-55, 556-8
inland waterway, OJ L 130 316–17	Preamble509
Recital 36317	Art 1509
Council Regulation No 2011/73, OJ 1973	Art 2509
L206/3509n	Art 2(b)
Council Regulation No 1411/77, OJ	Art 2(d)
1977 L160/4	Art 3
Council Regulation No 1681/79, OJ 1979	
L196/1509n	Art 3(1)
Council Regulation (EEC) No 2956/79 on	Art 3(1)(a)(i)
opening, allocating, and providing for	Art 3(1)(a)(ii)
the administration of a Community	Art 3(1)(a)(iii)
tariff quota for frozen beef and veal 206	Art 3(1)(a)(iv)
Council Regulation (EC) No 3017/79 amended by Council Regulation 1580/82, OJ 1979	Art 3(2)
L339/1	Art 3(2)(b)526n
Council Regulation (EC) No 2176/84 corrected by	Art 4509, 516
OJ 1984 L227/35 and amended by Council	Art 4(1)524, 556
Regulation 1761/87, OJ 1984 L201/1509	Art 4(2)
Council Regulation (EC) No 2423/88 corrected	Art 4(2)(a)
by OJ 1988 L264/58 and further amended	Art 4(2)(b)524, 527-8
by Council Regulations 521/94 and 522/94,	Art 4(2)(c)
OJ 1988 L209/1509	Art 4(3)
Regulation No 1893/91 of 20 June 1991 amending	Art 4(4)(a)
Regulation (EEC) No 1191/69 on action by	Art 4(4)(b)534
Member States concerning the obligations inherent in the concept of a public service	Art 5 509, 535–6
in transport by rail, road and inland	Art 6509, 536
waterway, OJ L 169170n	Art 7
Council Regulation (EEC) No 2834/91 of 23	Art 7(1)(a)
September 1991 collecting definitively the	Art 7(1)(b)
provisional countervailing duty on imports	Art 7(2)
of polyester fibres and polyester yarns	
originating in Turkey, OJ 1991 L272/92546	Art 7(3)
Council Regulation No 2408/92 of 23 July	Art 7(4)
1992 on access for Community air	Art 8
carriers to intra-Community air routes, OJ L 240	Art 8(1)
Art 4	Art 8(2)
Art 4(1)(i)	Art 8(3)
	Art 8(4)
Art 4(3)	Art 8(5)
Council Regulation (EEC) No 2913/92 of 12 October 1992 establishing the Community	Art 8(6)
Customs Code	Art 9510
Art 236 555	Art 9(1)

Art 10510	Annex I
Art 10(1)	Annex II
Art 10(2)	Annex III
Art 10(3)	Annex IV510
Art 10(6)	Council Regulation (EC) No 2026/97, OJ 1997
Art 10(7)	L288/1509
Art 10(8)	Council Regulation No 994/98 on the application
Art 10(9)	of Articles 92 and 93 of the Treaty establish-
Art 10(12)	ing the European community to certain categories of horizontal aid [1998]
Art 10(13)	OJ L142/1 (Enabling Regulation) 24, 31 <i>n</i> ,
Art 11510	158n, 314, 324-5
Art 11(1)557	Council Regulation (EC) No 1540/98 of
Art 11(2)	29 June 1998 establishing new rules on
Art 11(9)	aid to shipbuilding, OJ 1998 L 202/138 <i>n</i>
Art 12510, 557	Council Regulation No 659/1999 of 22 March
Art 12(1)	1999 on the application of Article 93 (now
Art 12(6)	Article 108) (Procedural Regulation) 23, 28–30, 45, 345
Art 13510	Art 130
Art 14510	Art 1(h)
Art 14(1)548n	Art 4(2)
Art 14(4)	Art 4(3)
Art 14(5)	Art 4(4)
Art 15510	Art 4(5)
Art 15(2)	Art 4(6)
Art 16510, 558	Art 7(6)
Art 16(4)	Art 10(1)
Art 17550	Art 10(3)
Art 18510, 558	Art 14
Art 18(1)	Art 18
Art 18(2)	Art 20(2)
Art 19510, 550, 558	447-8n
Art 20510, 557	Art 20(2)(1)
Art 21510, 550	Art 2230
Art 22510, 558	Art 24449
Art 23510, 549	Commission Regulation (EC) No 68/2001 of
Art 24510	12 January 2001 on the application of
Art 24(1)547-8	Articles 87 and 88 of the EC Treaty
Art 24(2)	to training aid
Art 24(5)	Commission Regulation (EC) No 69/2001 of
Art 26510	12 January 2001 on the application of
Art 27510	Articles 87 and 88 of the EC Treaty to <i>de minimis</i> aid [2001] OJ L10/30158, 464
Art 28510, 546-7, 554, 557	Commission Regulation (EC) No 70/2001 of
Art 29510, 557	12 January 2001 on the application of
Art 30510	Articles 87 and 88 of the EC Treaty to
Art 31510	State aid to SMEs, OJ L 10
Art 31(1)	Council Regulation (EC) No 1049/2001 of
Art 31(2)545n	30 May 2001 regarding public access
Art 31(3)545	to European Parliament, Council and Commission documents [2001]
Art 31(4)	OJ L145/43
Art 31(7)	Art 4(2)
Art 32556	Art 4(3)

Council Regulation (EC) No 1094/2001 of 5 June 2001 establishing the standard import values for determining the entry price of certain fruit and vegetables, OJ L 150	Commission Regulation (EC) No 1998/2006 of 15 December 2006 on the application of Articles 87 and 88 of the Treaty to <i>de</i> <i>minimis</i> aid, OJ L 379, 28.12.2006 32, 95,
Council Regulation (EC) No 1177/2002 of 27	158, 464
June 2002 concerning a temporary defensive	Art 2273n
mechanism to shipbuilding, OJ 2002	Art 2(2)
L 172/138n Council Regulation (EC) No 1407/2002 of 23 July 2002 on State aid to the coal	Regulation (EC) No 1370/2007 of the European Parliament and of the Council of 23 October
industry	2007 on public passenger transport services, OJ L315/1
Council Regulation No 1973/2002, OJ 2002	Art 3(1)
L305/4	Art 4
Commission Regulation (EC) No 2204/2002	Art 5
of 12 December 2002 on the application of	
Articles 87 and 88 of the EC Treaty to	Art 6
State aid for employment, OJ L 337324, 464	Art 7
Council Regulation (EC) No 1/2003 of 16	Art 9(1)
December 2002 on the implementation	Art 10(1)
of the rules on competition laid down in Articles 81 and 82 of the Treaty,	Commission Regulation (EC) No 800/2008
OJ 2003 C1/1	of 6 August 2008 on the application of
Art 9	Articles 87 and 88 (General Block Exemption Regulation) 24 <i>n</i> , 32, 34–5, 241, 324, 464
Council Regulation (EC) No 452/2003 of	Commission Regulation (EC) No 1008/2008
6 March 2003 on measures that the	on common rules for the operation of air
Community may take in relation to the	services in the Community, OJ L 293 308,
combined effect of anti-dumping or	319–20, 322–3
anti-subsidy measures with safeguard	Art 17319
measures, OJ 2003 L69/8 515	Council Regulation No 597/2009 last amended
Council Regulation (EC) No 1435/2003 of 22 July	by Regulation (EU) No 37/2014, OJ 2009
2003 on the Statute for a European Cooperative	L188/93
Society (SCE), OJ 2003 L 207	Council Regulation (EC) No 723/2009 of 25 June
Council Regulation No 461/2004, OJ 2004	2009 on the Community legal framework
L77/12	for a European Research Infrastructure
Commission Regulation (EC) No 794/2004 of 21 April 2004 implementing Council	Consortium (ERIC), OJ 2009, L206/1
Regulation (EC) No 659/1999 laying	Art 2(1)
down detailed rules for the application	Council Regulation No 1225/2009 of 30 November
of Article 93 of the EC Treaty,	2009 on protection against dumped imports from countries not members of the European
OJ L	Community (codified version), OJ 2009
Art 4328 <i>n</i>	L343/51
Art $6(1)$	Commission Regulation No 182/2011 of the
Commission Regulation (EC) No 1628/2006	European Parliament and of the Council
of 24 October 2006 on the application of	of 16 February 2011 laying down the rules
Articles 87 and 88 of the Treaty to national	and general principles concerning mecha-
regional investment aid granted by Member States	nisms for control by Member States of the Commission's exercise of implementing
Regulation (EC) No 1893/2006 of the European	powers, OJ 2011 L55/13515
Parliament and of the Council of 20 December	Commission Regulation No 360/2012 of
2006 establishing the statistical classification	25 April 2012 on the application of Articles
of economic activities (NACE Revision 2),	107 and 108 of the TFEU to de minimis aid
OJ L 393, 30.12.2006	granted to undertakings providing SGEI
Art 27.11247n	Text with EEA relevance, OJ 2012,
Art 27.20	L114/8
Art 29.10	Art 1(2)(g)
Art 29.32 247n	Art 2(2)

Commission Regulation No 966/2012 of 25	Art 89188
October 2012 on the financial rules applic-	Art 90196 <i>n</i>
able to the general budget of the Union and	Art 95189
repealing Regulation No 1605/2002,	Arts 101-2
OJ 2012, L298 (Euratom Regulation)199	Art 123(1)
Council Regulation (EC) No 733/2013 of	
22 July 2013 amending Regulation No 994/	Art 127
98 (Enabling Regulation)33-4, 287-8,	Art 128
324–5, 332	Annex X
Council Regulation (EC) No 734/2013 of	Annex XI
22 July 2013 amending Regulation	Council Regulation (EU) No 1305/2013 of
No 659/1999 29, 33–4, 44, 45, 47, 48 <i>n</i> , 52–3, 406–7, 411	17 December 2013 on support for rural
, ,	development by the EAFRD and repealing
Recital 1	Council Regulation (EC) No 1698/2005,
Recital 7367 <i>n</i>	OJ 20/12/2013 L 347/487 (Overarching ESI
Art 4(5)	Funds Regulation)211-12
Commission Regulation (EC) No 1224/2013	Council Regulation (EU) No 1306/2013 of
of 29 November 2013 amending Regulation	17 December 2013 on the financing, manage-
No 800/2008	ment, and monitoring of the common agri- cultural policy, OJ 20/12/2013 L 347/549
Commission Regulation (EC) No 1303/2013 of	
17 December 2013 laying down common	Recital 6
provisions on the ERDF, the ESF, the CF,	Regulation (EU) No 1307/2013 of 17 December
the EAFRD, and the EMFF, OJ L 347/320 20/	2013 establishing rules for direct payments to
12/2013	farmers under support schemes within the framework of the common agricultural
199–200, 203	policy211 <i>n</i>
Recital 10190	Regulation (EU) No 1308/2013 of 17 December
Recital 36	2013 establishing a common organization of
Recital 38	the markets in agricultural products OJ 20/
Recital 58203 <i>n</i>	12/2013 L 347/671 (CMO Regulation) 211 <i>n</i>
Recital 66	Recital 159212n, 216n
Recital 82194 <i>n</i>	Recital 160
Recital 83	Art 23
Recital 87	
Recitals 92-94	Art 219
Recital 108	Art 221216 <i>n</i>
	Council Regulation (EU) No 1379/2013 of
Art 2(13)	11 December 2013 on the common organization of the markets in fishery and
Art 4(3)	aquaculture products, OJ L354
Art 4(4)190 <i>n</i>	* *
Art 5(1) 189–90 <i>n</i> , 189 <i>n</i>	Recital 6
Art 9 188–9 <i>n</i>	Commission Regulation (EC) No 1407/2013
Art 37198	of 18 December 2013 on the application of Articles 107 and 108 to <i>de minimis</i> aid 33-4,
Art 37(1)	158-9, 290, 464
Art 37(2)	Recital 13210
Art 37(2)(b)	
Art 37(4)	Art 3
	Art 3(2)
Art 37(7)	Art 3(5)
Art 38198, 199	Art 6159
Art $38(4)(c)$	Council Regulation (EU) No 372/2014 of
Art 39199	9 April 2014 amending European
Art 61198	Commission Regulation 794/2004 [2014]
Art 73190 <i>n</i>	OJ L 109/14
A ## 75 100 cc	Art 11(b)

Council Regulation (EU) No 651/2014 of 17 June	Art 1154, 458
2014 declaring certain categories of aid	Art 11(1)
compatible with the internal market in appli-	Art 11(2)
cation of Articles 107 and 108 of the Treaty,	Art 12
OJ L 187/1 (GBER 2014) see under Other	Art 12(3)
Instruments	Art 13
Regulation (EU) No 654/2014 of the European Parliament and of the Council of 15 May	Art 14
2014 concerning the exercise of the Union's	Art 14(3)
rights for the application and enforcement	Art 15
of international trade rules and amending	Art 15(2)
Council Regulation (EC) No 3286/94,	Art 16
OJ 2014 L189/50508n	Art 16(1)
Council Regulation (EU) No 2015/1588 of	
13 July 2013 on the application of Articles 107 and 108 to horizontal State aid,	Art 16(2)
OJ 2015 L 248/1 48–9, 287–8, 451 <i>n</i> , 463	
Art 1	Art 17(3)
Council Regulation (EU) No 2015/1589 of	Art 18(1)
22 July 2013 replacing Regulation No 659/	Art 19
1999 (Procedural Regulation) 29, 45, 345,	Art 19(1)
348–53, 355–6, 361–8, 370–1, 451 <i>n</i>	Art 20
Preamble	Art 20(2)
Art 1 lit.(b)(ii)	Art 20(3)
Art 1 lit.(g)	Art 21
Art 1(h)	Art 22
Arts 2-11	Art 22(2)
Art 4	Art 23
Art 4(1)	Art 24
Art 4(2)	Art 24(2)
Art 4(3)	Art 24(2)(1)
Art 4(4)	Art 24(2)(2)
Art 4(5)	Art 25
Art 4(6)	Art 25(1)
Art 5(3)	Art 27
Art 6	
Art 7 366n, 370, 377, 379, 406–7, 409	Art 32(2)
Art 7(1)	Art 32(3)54
Art 7(2)	Decisions (in chronological order)
Art 7(6)	Commission Decision of 25 July1990, Italian aid
Art 7(7)	to assist certain areas of the Mezzogiorno
Art 8 366n, 377, 379, 406–7	affected by earthquakes
Art 8(1)	Commission Decision of 6 July 1994, for Case C-7/93
Art 8(2)	on reimbursement of the deficit sustained by TAP on the routes to the Atlantic islands,
Art 8(5)	OJ C178, 30/06/1993
Art 9	Commission Decision of 9 December 1998
Art 9(3)	regarding aid for natural disasters in the
Art 9(4)	Netherlands, NN 136/98, Official Journal C
Art 9(8)	72, 16/3/1999236 <i>n</i>
Art 9(9)	Commission Decision of 21 August 1999,
Art 9(10)	NN 70/98, Kinderkanel and Phoenix,
Art 10	OJ C 238303 <i>n</i> Commission Decision NN 394/2000,
Art 10(1)	OJ 2000 L 150
Art 10(3)	0,2000210011111111111111111111111111111

to aid having a social character in favour of	Commission Decision, Netherlands State aid N 451/2004 and N 542/2004 - Behoud financiële
certain categories of passengers on certain	reserves door ziekenfondsen en het risicover-
air routes	eveningssysteem
Commission Decision of 18 March 2000, NN 88/98 BB 24 hours, OJ C 78303 <i>n</i>	Commission Decision SA.12522 (Sernam2) [2005] OJ L 195
Commission Decision of 14 December 2000, in Case C-10/2000 (ex NN 112/99 and 141/99)—Germany; Aid for STAMAG Stahl-und Maschinenbau AG, Saxony (OJ 2005, C-270/44)	Commission Decision 2005/842/EC of 28 November 2005 on the application of Article 86(2) of the EC Treaty to State aid in the form of public service compensation granted to certain undertakings entrusted
Commission Decisions N 560/2001 and NN 17/ 02 National Heritage Fund for Brighton West Pier Trust, OJ 2002 C239	with the operation of SGEI, OJ 2005 C312/67
Commission Decision, Advantages granted to the local swimming pool of Dorsten (Germany) (Leisure pool Dorsten) [2001] OJ C 172 156n	Art 2.1(c)
Commission Decision 2001/673/EC of 28 March 2001 (<i>Lintra</i>) [2001] OJ L 236/3369n	Art 2.4
Commission Decision of 11 December 2001 on the tax measures for banks and banking foundations implemented by Italy,	Commission Decision of 10 December 2005, NN 123/2005, Cultural heritage scheme to promote tourism, OJ C 3143021
OJ L 184, 13.07.2002	Commission Decision NN 70/2006, Aid Scheme to cinema in Finland303n
recovery of illegal aid [2002] OJ L 247/27 374 Council Decision of 3 May 2002 on the granting	Commission Decision of 25 January 2006, in Case C-80/2006 (2006/743/EC)—On the State Aid
of a national aid by the authorities of the Italian Republic in favour of road transport	implemented by the Netherlands for AZ and AZ Vastgoed BV (OJ 2006, L 307/149) 378
undertakings, OJ L131/14	Commission Decision, Ireland State aid SA. 18879 - E 6/2006, Unlimited guarantee in favour of the Voluntary Health Insurance Board (VHI)
Kingdom of the Netherlands in favour of road transport undertakings, OJ L131/12	Commission Decision of 22 November 2006, N 461/2005 (UK), UK film tax incentive
Council Decision of 3 May 2002 on the granting of aid by the French Government for road transport undertakings,	Commission Decision 2007/374 of 24 January 2007 in Case C 52/2005
OJ L131/15	Commission Decision SA.20674 (South Yorkshire Digital Region Broadband Project) [2007] OJ C 80/2681
N588/02, United Kingdom - Grant for long- distance coach services	Commission Decision of 27 June 2007 State aid C 50/2006 (ex NN 68/2006, CP 102/2006)
Recitals 47-49	implemented by Austria for BAWAG-PSK, OJ L 83, 26.03.2008
the aid scheme implemented by Belgium in favour of coordination centres established in Belgium, C(2003) 564 final353n	Commission Decision of 5 September 2007, N 276/2007, Aid measures with a
Commission Decision of 24 April 2003, NN 136a/ 2002, Ecomusée d'Alsace, OJ C 96 302n	cultural objective under the Regional Development Operational Programmes, OJ C 206/20073021
Commission Decision, Ireland State aid N 46/2003 - Risk equalization scheme in the Irish health insurance market102, 354	Commission Decision of 28 November 2007, N 605/2007 (Malta), Financial incentives for
Commission Decision, Advantages granted to a	the Audiovisual Industry Regulations305 <i>n</i>
conference centre in Sweden (Aid in favour of a congress hall in Visby—Gotland) [2003] OJ C 75	Commission Decision of 5 December 2007, case No NN 70/2007—United Kingdom— Northern Rock, OJ C 43, 16.02.2008 2688
Commission Decision of 12 December 2003, NN 221/2003, Aviodrome, OJ C 301302n	Commission Decision of 19 December 2007, NN 50/2007, State guarantee in favour of

Commission Decision, Lithuania State aid Case N 764/2007, Construction of a 400 MW Combined Cycle Gas Turbine Plant at AB	Republic of Poland for the purchase of agricultural land between 1 January 2010 and 31 December 2013, OJ L4/89
Lietuvos Elektrinė, C(2008) 457 final 205–6	Commission Decision of 2 December 2009, in
Commission Decision, Czech Republic, N627/2007 C (2008) 2605 on Investment aid scheme for the reduction of VOC emissions 207–8	Case C-36/2008 (ex NN33/2008)—Italy; Sulcis power plant project
Commission Decision of 8 February 2008, NN 377/2007, Support to Bataviawerf—	Council Decision of 16 December 2009 on the granting of State aid by the authorities of the Republic of Latvia for the purchase of
Reconstruction of a vessel from the 17th century, OJ C 35/2008302n	agricultural land between 1 January 2010 and 31 December 2013, OJ L339/34223
Commission Decision of 2 April 2008, N 562/ 07 (Spain), Ayudas a la créaciondesarrollo y producción audiovisual en el País Vasco 305 <i>n</i>	Council Decision of 16 December 2009 on the granting of State aid by the authorities of the Republic of Lithuania for the purchase of
Commission Decision of 16 July 2008, in Case C-18/2008 (ex NN 12/2008)—Hungary; Dunaferr378	state-owned agricultural land between 1 January 2010 and 31 December 2013, OJ L338/93
Commission Decision of 16 July 2008, N 60/ 2008 (Italy), Sostegno al cinema della Regione Sardegna	Council Decision of 22 December 2009 on the granting of State aid by the authorities of the Republic of Hungary for the purchase of
Commission Decision of 16 July 2008, N 202/ 08 (Hungary), Hungarian film support	agricultural land between 1 January 2010 and 31 December 2013, OJ L348/55 223
schemes	Commission Decision, Romania State aid N 467/ 2009—State aid scheme for supporting investments in expanding and modernizing distribution networks for electricity and
Commission Decision, Ireland State aid N 582/ 2008 - Health Insurance intergenerational solidarity relief	natural gas
Commission Decision 2008/92/EC (Aid to the Sardinian Shipping Sector) [2008] OJ L29/24	contracts between the Danish Ministry of Transport and Danske Statsbaner, OJ 2011 L 7
Commission Decision of 10 December 2008, N 477/2008 (Germany), German film support scheme	Commission Decision N 178/2010 of 29 September 2010, Spain—Public service compensation linked to a preferential dispatch mechanism for indigenous coal power plants
Commission Decision of 2 July 2009 N 106/2009 (France), Crédit d'impôt pour les oeuvres cinématographiques étrangères	Commission Decision of 8 December 2010, Ordre national des pharmaciens en France (ONP), Case COMP/39.510
Commission Decision of 13 August 2009, Homeowners mortgage support scheme, OJ C 209 2	Council Decision 2101/787 of 10 December 2010 on State aid to facilitate the closure of uncompetitive coal mines, OJ L 336,
Commission Decision of 24 November 2009, Guarantee by government for housing loans, OJ C 25 2	21.12.2010
Commission Decision N 459/A/2009 - Italy, Aid scheme for compensation for damage caused by the earthquake in Abruzzo of	Netherlands: Aid for a CCS project in the Rotterdam harbour area, C (2010) 7254 final209
6 April 2009	Commission Decision 2011/501/EU of 23 February 2011 on Aid for the Bahnen der
State aid C-41/05 awarded by Hungary through Power Purchase Agreements [2009] OJ L 225	Stadt Monheim (BSM) and Rheinische Bahngesellschaft (RBM) companies in the Verkehrsverbund Rhein Ruhr,
Commission Decision NN 33/2009 (Belgium),	OJ 2011 L 210
Régime d'aides en faveur de la production d'oeuvres audiovisuelles	2011 on State aid measure No C 24/2009 (ex N 446/2008) State aid for energy-intensive
Council Decision of 20 November 2009 on the	businesses under the Green Electricity Act in Austria, OI 2011 L 235/42

Commission Decision of 13 July 2011 SA. 30596 (N 101/2010) DE—LIP- GLOBALFOUNDRIES Group (Fab Booster Investment and Fab 1 Annex), Dresden, OJ C 258	Apple - Invitation to submit comments pursuant to Article 108(2) of the Treaty on the Functioning of the European Union, C 369/22
Commission Decision SA.32163 of 1 August 2011, Remediation of damage to airlines and airports caused by seismic activity in Iceland and the volcanic ash in April 2010 237 Commission Decision of 19 October 2011, SA.32612 Restructuring Aid to PKS Pulawy Sp	Commission Decision, Netherlands State aid SA.38374 (2014/C) (ex 2014/NN) (ex 2014/CP) - Alleged aid to Starbucks - Invitation to submit comments pursuant to Article 108(2) of the Treaty on the Functioning of the European Union, C 460/11
Recital 5	Commission Decision of 10 December 2014, SA.38751 (2014/N)—Czech Republic— Evaluation plan for the block exempted large aid scheme 'Law on investment incentives,' C (2014) 9931 final
Recital 5	Commission Decision of 11 December 2014, SA.39273 (2014/N)—United Kingdom, Evaluation plan for the block exempted large aid scheme 'Regional Growth Fund,' C (2014) 9319 final
TFEU to State aid in the form of public service compensation granted to certain undertakings entrusted with the operation of SGEI, OJ 2012, L7/3 23, 86, 90, 95, 97, 100, 318–19, 320 Art 2.1(d)	United Kingdom—Alleged State aid to Glenmore Lodge
Art 3	Commission Decision SA.38208 of 29 April 2015, United Kingdom—Alleged State aid to UK member-owned golf clubs JOCE C/277/2015
form of public service compensation granted to certain undertakings entrusted with the operation of services of general economic interest (SGEI Decision)	Notices (in chronological order) Commission Notice 97/C 238/02 on Community guidelines on State aid for rescuing and restructuring firms in difficulty (OJ 1997 C 283)
London) [2012] OJ C 220/6	Commission Notice 98/C 384/03 on the application of the State aid rules to measures relating to direct business taxation, OJ C 384, 10 December 1998 0003-0009 69n, 133, 167n
Commission Decision on the State Aid Scheme N SA.36139, which the UK is planning to implement for video games, C(2014) 1786 final	Commission Notice on the application of Articles 87 and 88 of the EC Treaty to State aid in the form of guarantees, OJ C 071, 11 March 2000 0014-0018 167n
Commission Decision 2014/99/EU of 18 February 2014 setting out the list of regions eligible for funding from the ERDF and the ESF and of Member States eligible for funding	Commission Notice on the determination of the applicable rules for the assessment of unlawful State aid, OJ 2002 C119/22 227
from the CF for the period 2014–20, OJ L 50/22 20/02/2014	Commission Notice on the application of Articles 87 and 88 of the EC Treaty to State aid in the form of guarantees, OJ C 155, 20 June 2008
NGN srl	Commission Notice on the enforcement of State aid law by national courts, OJ 2009 C 85/1 s 3

§ 78(e)335 <i>n</i>
EU Guidelines for the application of State aid rules in
relation to the rapid deployment of broadband
networks, OJ 2013, C25/01 98 <i>n</i> , 103-4, 243
Commission Guidelines on regional State
aid for 2014-20, 23/07/2013 C 209/1
('RAG 2014–20')194–6, 241, 241 <i>n</i> ,
243-5, 246-7, 252-4, 256, 257-64
Recital 5
Recital 96
Recital 97245 <i>n</i>
Recital 113241n
Recital 146
Recital 183
§ 96246 <i>n</i>
§ 117
§ 145195
§ 146195
§ 151195
Annex II
Annex IV
fn 14
Communication from the Commission,
amending other Communications
from the Commission on EU Guidelines
for the application of State aid rules
COM(2014) 3349/2
Guidelines on State aid to airports and airlines,
Communication from the Commission
OJ C99/03, 4 April 2014 (Aviation
Guidelines)
Guidelines on State aid for environmental
protection and energy 2014-2020,
Communication from the Commission
OJ C 200, 28 June 2014
Commission Guidelines on State aid for rescuing and restructuring non-financial
undertakings in difficulty, OJ C 249,
31 July 2014 (Rescue and Restructuring
Guidelines)
Other Instruments
Communication from the Commission
OJ 1983, C 318/3382
Communication from the Commission to the
Member States concerning public
Member States concerning public authorities' holdings in company capital
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)173 <i>n</i>
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)173 <i>n</i> Community Framework for State aid for
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)173 <i>n</i> Community Framework for State aid for research and development, OJ C 83,
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)
Member States concerning public authorities' holdings in company capital (Bulletin EC 9-1984, pt 3.5.1)

Article 5 of Commission Directive 80/723/ EEC to public undertakings in the manufacturing sector (OJ 1993 C 307)	Community Framework for State aid for research and development and innovation, OJ C 323, 30 December 2006
Communication from the Commission 94/C 350/07 entitled 'Application of Articles 92 and 93 of the EC Treaty and Article 61 of the EEA Agreement to State Aids in the Aviation Sector' (OJ 1994 C 350)	Communication from the Commission, A single market for 21st century Europe—Services of general interest, including social services of general interest: a new European commitment, COM(2007) 725
research and development, OJ C 45, 17 February 1996281	Communication from the Commission on the application of State aid rules to measures
Communication from the Commission on State aid elements in sales of land and buildings by public authorities, OJ 1997 C 209/3	taken in relation to financial institutions in the context of the current global financial crisis, OJ, C 270, 25 October 2008 (Banking Communication) 266, 269–71, 274, 279
Communication from the Commission on the Links between Regional and Competition	§ 2.11 .270 § 21 .274
Policy, OJ C90, 26/03/1998	§ 51269
Communication from the Commission on the application of Articles 92 and 93 of the EC Treaty and Article 61 of the EEA Agreement to State aid in the aviation sector, OJ C 350, 10 December 1999	Communication from the Commission: The Recapitalization of financial institutions in the current financial crisis: Limitation of the aid to the minimum necessary and safeguards against undue distortions of competition,
Commission Interpretative Communication on	OJ C 10, 15 January 2009 (Recapitalization
the Community law applicable to public	Communication)
procurement and the possibilities for integrating environmental considerations into public procurement, COM(2001) 274, 4 July 2001	Communication from the Commission, Temporary framework for State aid measures to support access to finance in the current financial and economic crisis, OJ C 16,
Commission Interpretative Communication on the Community law applicable to public	22 January 2009
procurement and the possibilities for inte- grating social considerations into public procurement, COM(2001) 566,	State aid measures to support access to finance in the current financial and economic crisis, 2009/C 83/01209n
15 October 2001180 <i>n</i>	Commission Code of Best Practice for the
Communication from the Commission on the application of State aid Rules to public service broadcasting, OJ C 320, 15 November 2001	conduct of State aid control procedures [2009] OJ C 136/13 (Best Practice Code)
Communication from the Commission	Recital 48
concerning State aid N 376/01—Aid scheme	§ 11
for cableways—Authorization of State aid	§ 15356
under Articles 87 and 88 of the EC Treaty, OJ 2002 C 172/2	Communication from the Commission on the treatment of impaired assets in
_	the community banking sector, OJ C
Recital 16	72, 26 March 2009 (Impaired Assets
	Communication)
Recital 39	Communication from the Commission on the
Communication from the Commission on certain legal aspects relating to cinematographic and	return to viability and the assessment of re- structuring measures in the financial sector
other audio-visual works, C 43, 16 February 2002 (Cinema Communication) 305–6	in the current crisis under the State aid rules,
Communication from the Commission on the	OJ C 195, 19 August 2009 (Restructuring Communication)266, 272
promotion of cooperative societies in	Communication from the Commission
Europe COM(2004) 18 final	concerning the criteria for an in-depth
Community Framework for State aid in the form	assessment of regional aid to large
of public service compensation, OJ 2005	investment projects (IDAC), OJ C 223,
C297/0495-6	16 September 2009

Communication from the Commission, Europe 2020, A strategy for smart, sustainable and inclusive growth, 03 March 2010, COM(2010) 2020 final	§ 20. .283n § 28. .285n § 29. .285n § 32. .285n § 33. .285n § 38. .288 § 104. .288 fn 22. .283n Community Framework for State aid for research
for Europe 2020' COM(2011) 500 final 187 <i>n</i> Communication from the Commission, A Quality	and development and innovation, OJ C 198, 27 June 2014
Framework for Services of General Interest	288-9, 290, 292-7, 300
in Europe, Brussels, 20 December 2011,	Art 1(2)(a)
COM(2011) 900 final97	Art 1(4)(a)
Communication from the Commission on an EU	Art 1(5)
framework for State aid in the form of public service compensation, OJ 2012, C8/15	Art 2(1)(43)
(SGEI Framework)	Art 2(1)(44)
\$ 2.2	Art 2(1)(a)
§ 2.3	Art 2(1)(b)
§ 2.4	Art 2(18)
§ 2.5	Art 6(2)
§ 2.6	Art 8(1)
§ 2.8	Art 8(2)
§ 11	Art 9
§ 16	Art 13
Recital 8	Art 14
Recitals 39-43	Art 14(7)
Communication from the Commission on the	Art 15
application of the European Union State aid	Art 21
rules to compensation granted for the	Art 21(13)
provision of services of general economic	Art 22
interest, OJ 2012, C8/4 74, 75, 80–1, 86,	Art 26
90–1 <i>n</i> , 97	Art 31
Communication from the Commission on EU State Aid Modernization (State Aid Action	Art 32
Plan), COM(2012) 209 final 231–3, 281,	Art 33
324, 345, 361	Art 34
European Parliament Resolution of 17 January 2013	Art 35
on State Aid Modernization (2012/2920(RSP))	Art 41
§ 835	Art 44
§ 1335n	Art 46
§ 14	Art 48
Communication from the Commission on State	Art 50
Aid for Films and Other Audio-Visual	Art 51
Works, OJ C 332, 15 November 2013 (Cinema Communication II)305–7	Art 52
Council Regulation (EU) No 651/2014 of	Art 53 307 <i>n</i> , 327, 332, 335–6
17 June 2014 declaring certain categories of	Art 54
aid compatible with the internal market in	Art 55
application of Articles 107 and 108 of the	Art 56
Treaty, OJ L 187/1 (GBER 2014) 194, 203,	Art 58(4)
220, 237, 244–5, 252–7, 288–9, 291, 300, 307, 324–38, 397–8, 464	Annex II
\$ 19	Annex III
y 17	

	150 100 101
Rules of Procedure of the General Court 1991	Art 6
Art 44(1) lit. c	Art 6.1
Art 48(2)	Art 6.3
Art 113	Art 6.9
Rules of Procedure of the General Court 2015	Art 7501n
Art 76 lit. c	Art 8494
Art 84(1)	Art 8.2
Art 84(2)	Art 9494
Art 129432	Art 10501 <i>n</i>
	Art 11.1
NATIONAL LEGISLATION	Art 11.2
NATIONAL LEGISLATION	Art 11.4
Italy	Art 11.6
Civil Code	Art 11.7
Art 2909	Art 11.8
c ·	Art 11.9
Spain	Art 11.11557
Constitution	Art 12.1
Art 105(c)344n	Art 12.1.2511, 557
United Kingdom	Art 12.1.3498
Single European Act 1986	Art 12.2
ongle Baropean rec 190021	Art 12.5
United States	Art 12.6
Constitution	Art 12.7
Fifth Amendment182 <i>n</i>	Art 12.8
	Art 12.9
	Art 13498
INTERNATIONAL	Art 14
INSTRUMENTS	Art 15.1
World Trade Organization	Art 15.2
Agreement on Subsidies and Countervailing	Art 15.4
Measures 1994 (SCM Agreement) 20n, 36,	Art 15.5
467, 471, 474–99, 500, 501–2,	Art 15.7
508, 509, 556-8	Art 17.1
Art 1	Art 17.2
· · · · · · · · · · · · · · · · · · ·	Art 17.4
Art 1.1(a)(1)	Art 18
Art 1.1(a)(1)(i)	Art 18.6
Art 1.1(a)(1)(ii)	Art 19.3
Art 1.1(a)(1)(iii)	Art 20
Art 1.1(a)(1)(iv)	Art 21.1
Art 1.1(a)(2)	Art 21.2
Art 1.2	Art 21.3
Art 2	Art 22
Art 2.1	Art 23
Art 2.1(c)	Art 24.2
Art 3	Art 24.3
Art 3.1	Art 24.4
Art 3.2	Art 24.5
Art 4.7	Art 25.10
Art 5	
	Art 27.10

Agreement on Subsidies and Countervailing	Art 21(1)501
Measures 1994 (SCM Agreement) (cont.):	Annex 1500
Annex I	Annex 2213, 501
fn 35	Annex 3500
fn 36495 <i>n</i>	Anti-Dumping Agreement 1994496
fn 46497	General Agreement on Tariffs and Trade
Agriculture Agreement 1994 212–16, 471, 500–2	(GATT) 1947/199419, 36, 467, 469–73
Art 3500	Art II
Art 3(3)	Art II(2)(b)
Art 6213, 500	Art III470, 497
Art 6(2)501	Art III(2)
Art 6(4)501	Art III(8)(b)470
Art 6(5)	Art VI470, 493, 495
Art 7501	Art XVI
Art 9213	Art XX494
Art 9(1) 500, 500 <i>n</i>	Art XX(g)502
Art 9(1)(a)	Art XXI
Art 9(1)(b)213	Art XXIII472, 493
Art 9(1)(c)	General Agreement on Trade in Services
Art 9(1)(d)213	(GATS) 1994
Art 9(1)(e)	Art III:8
Art 9(1)(f)	Art VI
Art 10500	Art XVI:2-5
Art 13501, 501 <i>n</i>	Art XVII
Art 15501	WTO Dispute Settlement Understanding
A rt 21 2124	(DSU)

List of Abbreviations

AAS Advanced Authorization Scheme

AG Agriculture Agreement

AMS Aggregate Measurement of Support

AS Anti-subsidy

BOPT business property occupation tax CAP Common Agricultural Policy

CEPIFINE Confederation of European Fine Paper Industries

CF Cohesion Fund

CFI Court of First Instance
CFP Common Fisheries Policy
CFR Charter of Fundamental Rights
CJEU Court of Justice of the European Union

CMEs coordinated market economies

CVDs countervailing duties

DAC Development Assistance Committee

DDS Duty Drawback Scheme

DEPBS Duty Entitlement Passbook Scheme

DSB Dispute Settlement Body

DSU Dispute Settlement Understanding

EAFRD European Agricultural Fund for Rural Development

EBIT earnings before interest and taxes

EBRD European Bank for Reconstruction and Development

EC European Community Treaty

ECHR European Convention on Human Rights

ECJ European Court of Justice

ECSC European Coal and Steel Community
EEA European Economic Area Agreement
EEC Treaty European Economic Community Treaty
EEPR European Energy Programme for Recovery
EMFF European Maritime and Fisheries Fund

EOUs Exported Oriented Units

ERDF European Regional Development Fund

ESF European Social Fund
FI Financial Instruments
FMS Focus Market Scheme
FOS Farmers' Sales Organization

GATS General Agreement on Trade in Services
GATT General Agreement on Tariffs and Trade
GBER General Block Exemption Regulation

GC General Court

HHI Herfindahl-Hirschman Index IDAC In-depth assessment communication

IIDSF Ignalia International Decommissioning Support Fund

IIP injury investigation period
IP investigation period
IPRs intellectual property rights
IRR internal rate of return
KDB Korea Development Bank

KEIC Korean Export Insurance Corporation

LDR Less Developed Regions

LMEs liberal market-oriented economies

MA Managing Authorities
MDR More Developed Regions

MEIP market economy investor principle MEIT market economy investor test

MfN most favoured nation

NBER National Bureau of Economic Research NGO Non-Governmental Organization

NHS national health services

NOS Nederlandse Omroep Stichting

NPV net present value

NUTS Nomenclature of Territorial Units for Statistics NVNI non-violation nullification and impairment

OECD Organization for Economic Cooperation and Development PACE partnership in assisting community expansion fund

PAT Autonomous Province of Trento
PPPs Private Public Partnerships
PSO public service obligations
R&D research and development
RAG Regional Aid Guidelines
RBI Reserve Bank of India

RDI Research, Development, and Innovation

SAAP State Aid Action Plan SAM State Aid Modernization

SCM Agreement on Subsidies and Countervailing Measures

SEZs Special Economic Zones

SGEI services of general economic interest

SGI services of general interest

SMEs small and medium-sized enterprises

SND Norwegian Industrial and Regional Development Fund

SOEs state-owned enterprises TCD Trinity College Dublin

TFEU Treaty on the Functioning of the European Union

TR Transition Regions
VoC variety of capitalism
WTO World Trade Organization
WWAN Wireless wide area networking

List of Contributors

Christopher Bovis (University of Hull)

Professor Christopher Bovis JD Phil, LLM, FRSA is Professor of Business Law at the University of Hull, UK. Bovis is a leading authority in public procurement and public–private partnerships and specializes in anti-trust law and policy, with particular emphasis on public sector management. He is Special Advisor to the UN on Public–Private Partnerships and Academic Advisor to the EU Commission on the Single Market. He advises national governments and industry across the world on public sector reforms and has been instrumental in structuring flagship public–private partnerships in the UK, EU, and overseas. He has acted on behalf of governments, the defence sector, and industry on numerous high-profile projects. He is Editor-in-Chief of *European Procurement and Public Private Partnerships Law Review* (EPPPL) published by Lexxion. His academic work has been translated into different languages, including Chinese, French, and German.

Christopher Bovis is co-author of Part II: The Notion of State Aid.

Raj Chari (Trinity College Dublin)

Raj Chari is an Associate Professor in Political Science, Trinity College Dublin (TCD), having earned his PhD from Queen's University (Canada). Chari's research has been broadly centred on comparative public policy, with a focus on competition policy (State aid control as well as Mergers and Acquisitions), the regulation of lobbyists from a global comparative perspective, and the role of different policymaking actors at both the domestic and EU levels of governance. Articles have appeared in high-ranking peer-reviewed journals such as *Government and Opposition, Electoral Studies, Political Quarterly, West European Politics*, and the *Journal of European Public Policy*. His previous books have been published by Manchester University Press as well as Routledge, and he has recently published a research monograph with OUP examining the life of privatized firms in France, Germany, Italy, Spain, and the UK (www.lifeafterprivatization.com).

Raj Chari is co-author of Part I: Rationales for State Aid Rules.

Roberto Cisotta (LUMSA University, Rome)

Roberto Cisotta works as Lecturer and Adjunct Professor of International and EU Law at LUMSA University (Rome) and as EU Law Expert at the Research Centre of the Italian Constitutional Court. In 2013/14 he worked as Contract Professor of International Law at the University of Viterbo 'La Tuscia'. He is also a qualified lawyer at the Bar of Rome. After earning his JD Degree at 'La Sapienza', University in Rome in 2005, he entered the PhD course in EU Law at the University of Trieste in consortium with the University of Rome Tor Vergata, which he attended while working as Teaching Assistant and which he completed successfully in 2009. His PhD thesis has been awarded the Riccardo Monaco prize by the Italian Society for International Law as the best PhD thesis in EU Law in Italy for 2009. From 2008 to 2010 he worked as Contract Professor at LUISS Guido Carli University in Rome. Between 2011 and 2012 he worked as référendaire at the General Court of the EU and then once more as Contract Professor at LUISS Guido Carli University in Rome. Since 2006 he has authored several major articles and essays in International and EU Law.

Roberto Cisotta is co-author of Part II: The Notion of State Aid.

Leigh Hancher (University of Tilburg; Allen & Overy)

Leigh Hancher is Professor of European Law at Tilburg University, NL, and specialises primarily in the field of interconnection between competition law and sector-specific regulation (especially energy law), and State aid, and cross-subsidization. She is Of Counsel with Allen & Overy in Amsterdam and is also Professor of the European Energy Law and Policy at the Florence School of Regulation, EUI, Italy. She is author of the leading handbook on *European State Aids* (4th edition, 2012), as well as numerous articles and books on competition law and regulated industries. She is a member of the editorial board of *European State Aids Law Quarterly*.

Leigh Hancher is co-author of Part IV: Procedures before the Commission.

Herwig C H Hofmann (University of Luxembourg)

Herwig C H Hofmann is Professor of European and Transnational Public Law and Jean-Monnet Chair in European Public Law at the University of Luxembourg. He researches, teaches, publishes, and litigates in European and transnational constitutional, administrative, and regulatory law as well as international economic law. A graduate of the Universities of Hamburg and Oxford, he formerly was lecturer in law at Trinity College in Dublin, and has been visiting professor and scholar at various universities including Sciences Po in Paris, Columbia Law School in New York, the University of San Francisco Law School, and Chuo University in Tokyo.

Herwig C H Hofmann is editor of this book and co-author of Part I: Rationales for State Aid Rules.

Ramona Ianus (European Commission, DG Competition)

Ramona Ianus studied law at the University of Bucharest and Pantheon Sorbonne Paris I and holds a Master of European Law from the College of Europe (Bruges). After obtaining the Master degree and qualifying to the Bar in Bucharest, she started her career in 2007 as the Directorate General for Competition in the department for regional state aid. In the first three years she dealt with a variety of cases, especially with regional aid granted to large investment projects. Thereafter, she was appointed to manage the real economy crisis aid measures. Having accumulated significant experience in the regional aid field, she was involved in the State Aid Modernization Package announced by the Commission in April 2012 and is responsible for the preparation of the regional aid rules for 2014–20.

Ramona Ianus is co-author of Part III: Compatibility Rules on State Aid.

Marc Jaeger (President of the General Court of the European Union)

Marc Jaeger has been the President of the General Court since September 2007 (ongoing third mandate). Previously, he was a judge at the Luxembourg District Court (1984), a judge at the Court of First Instance of the European Communities from July 1996, and then a President of Chamber (2004–7). He lectured at the universities of Luxembourg, Nancy II, and Rome (Luiss). He is a member of many scientific committees and reviews, and is the author of numerous contributions found in edited books, law journals, and other series.

Marc Jaeger has written the foreword of this book.

Viktor Kreuschitz (Judge at the General Court of the European Union)

Viktor Kreuschitz has been Judge at the General Court of the European Union since September 2013. Previously, he worked as a civil servant in the Federal Chancellery in the Constitutional Affairs Department (1981–97) and advisor in the Legal Service of the European Commission (1997–2013). Kreuschitz graduated as Doctor of Laws of the University of Vienna in 1981.

Viktor Kreuschitz is co-author of Part V: Judicial Review before the Court of Justice of the European Union.

Tim Maxian Rusche (European Commission, Legal Service)

Tim Maxian Rusche is, since 2010, a Member of the Legal Service of the European Commission. Prior to that, he worked in the Commission's Directorate General for Energy and Transport, first in the unit in charge of competition and internal market, and then as coordinator for relations with the European Parliament and the Council. He holds a joined law degree from University Paris 1 (Panthéon-Sorbonne) and the University of Cologne, a Master in Public Administration from the Kennedy School of Government (Harvard), and a doctorate in law from University Paris 1 (Panthéon-Sorbonne). He is the author of *EU Renewable Electricity Law and Policy* (Cambridge University Press), and has published several articles and book chapters on EU law, with a focus on State aid, regulation of energy and transport, and environmental law, as well as on other topics, such as leadership and foreign policy.

Tim Maxian Rusche is co-author of Part III: Compatibility Rules on State Aid.

Claire Micheau (European Commission, DG Competition; University of Helsinki)

Claire Micheau is a Policy Officer at the European Commission (DG Competition). She has also the title of Associate Professor at the University of Helsinki. She is a regular invited lecturer in several universities, including the University of Luxembourg, University Paris 1 Panthéon-Sorbonne,

and Leiden University. Previously, she held a teaching position in the law departments of Trinity College Dublin and Paris 1 Panthéon-Sorbonne. She graduated with a doctorate in law (Sorbonne Law School and University of Luxembourg, with Summa Cum Laude; PhD award *Prix Solennel de la Chancellerie des Universités de Paris* for the best doctoral thesis, and *Award for Outstanding PhD Thesis* of the Luxembourgish National Research Fund) and LLM at the University of Cologne. She is also qualified as an attorney at the Paris Bar. She has published various books and articles in the areas of EU law, WTO law, State aid law, and tax law.

Claire Micheau is editor of this book and co-author of Part I: Rationales for State Aid Rules.

Bucura C Mihaescu-Evans (University of Luxembourg)

Bucura C Mihaescu-Evans works as a Post-Doctoral Research Associate at the University of Luxembourg. She graduated in EU Law and Political Science from universities in France, England, and Luxembourg. She has recently defended her PhD as a fellow of the Marie Curie INCOOP Program of the European Commission (on the topic 'The right to good administration at the cross-roads of the various sources of fundamental rights in the EU integrated administrative system') for which she was awarded the best distinction. Previously, she worked for three years in the cabinet of Advocate General Yves Bot at the Court of Justice of the European Union. She is the author of several articles published in Romanian and foreign journals.

Bucura C Mihaescu-Evans is a member of the editing team of this book and co-author of Part IV: Procedures before the Commission.

Hanns Peter Nehl (Court of Justice of the European Union)

Hanns Peter Nehl has worked as a Legal Secretary in the cabinet of the Austrian Judges Josef Azizi and Viktor Kreuschitz at the General Court of the European Union since 1 September 2004. Previously, he worked as a Legal Officer in the Media Unit of Directorate-General for Competition of the European Commission (Media unit) as well as in the Brussels office of former German law firm Gaedertz Rechtsanwälte. He obtained his doctoral degree in 2001 at the University of Hamburg following research at the Graduiertenkolleg Integrationsforschung of the Europa-Kolleg Hamburg. He is also an alumnus of the European University Institute in Florence (LLM, academic year 1996/7) as well as of the Université d'Aix-Marseille III (D.E.A. de droit communautaire, academic year 1991/2). His main fields of interest are European competition law and European constitutional and administrative law in general. Main publications: *Principles of Administrative Procedure in EC Law* (Oxford: Hart Publishing, 1999; Europäisches Verwaltungsverfahren und Gemeinschaftsverfassung, Berlin: Duncker & Humblot, 2002).

Hanns Peter Nehl is co-author of Part V: Judicial Review before the Court of Justice of the European Union.

Aindrias Ó Caoimh (Judge at the Court of Justice of the European Union)

Aindrias Ó Caoimh has been Judge of the Court of Justice of the European Union since 13 October 2004; he was also Senior Counsel (1994–9), Judge of the High Court (Ireland) (1999–2004), and Barrister (Bar of Ireland) (1972–99).

Aindrias Ó Caoimh is co-author of Part II: The Notion of State Aid.

Massimo Francesco Orzan (Court of Justice of the European Union)

Dr Massimo Francesco Orzan is Legal Secretary of the President of the EU General Court. He graduated in Political Science at Luiss Guido Carli University (Rome) and in Law at University of Cassino. He holds a PhD in International Law and EU Law from Sapienza University (Rome). He has been Fellow of the Italian Ministry of Agricultural, Food, and Forestry Policies, and Post-Doc Fellow in EU Law, Assistant Professor of EU Law, and Adjunct Professor of International Economic Law and International Environmental Law at Luiss University and of International Law at Italian Navy Academy (Leghorn). He has also been speaker at conferences and workshops. Dr Orzan is author of numerous articles published in Italian and foreign journals. He has also been admitted to the Bar of Rome.

Massimo Francesco Orzan is co-author of Part III: Compatibility Rules on State Aid.